

JAMES Journal of Advances in Modeling Earth Systems*



10.1029/2021 MS002752

Key Points:

- A biogeochemical model was improved to simulate microbedriven soil organic matter (SOM) decomposition and carbon dynamics in agricultural ecosystems
- The model simulates carbon dynamics by simulating both microbial controls and impacts of soil, crop, and farming management practices
- Carbon input drives microbial activity and exerts large impacts on carbon dynamics while microbes play a central role in decomposing SOM

Correspondence to:

J. Deng, dengjia85@gmail.com

Citation:

Deng, J., Frolking, S., Bajgain, R., Cornell, C. R., Wagle, P., Xiao, X., et al. (2021). Improving a biogeochemical model to simulate microbial-mediated carbon dynamics in agricultural ecosystems. *Journal of Advances in Modeling Earth Systems*, *13*, e2021MS002752. https://doi. org/10.1029/2021MS002752

Received 13 AUG 2021 Accepted 28 OCT 2021

© 2021 The Authors. Journal of Advances in Modeling Earth Systems published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Improving a Biogeochemical Model to Simulate Microbial-Mediated Carbon Dynamics in Agricultural Ecosystems

Jia Deng¹, Steve Frolking¹, Rajen Bajgain², Carolyn R. Cornell^{2,3}, Pradeep Wagle⁴, Xiangming Xiao², Jizhong Zhou^{2,3,5}, Jeffrey Basara^{5,6}, Jean Steiner^{4,7}, and Changsheng Li¹

¹Earth Systems Research Center, Institute for the Study of Earth, Oceans and Space, University of New Hampshire, Durham, NH, USA, ²Department of Microbiology and Plant Biology, University of Oklahoma, Norman, OK, USA, ³Institute for Environmental Genomics, University of Oklahoma, Norman, OK, USA, ⁴USDA, Agricultural Research Service, Grazinglands Research Laboratory, El Reno, OK, USA, ⁵School of Civil Engineering and Environmental Science, University of Oklahoma, Norman, OK, USA, ⁶School of Meteorology, University of Oklahoma, Norman, OK, USA, ⁷Department of Agronomy, Kansas State University, Manhattan, KS, USA

Abstract Soil microbes drive decomposition of soil organic matter (SOM) and regulate soil carbon (C) dynamics. Process-based models have been developed to quantify changes in soil organic carbon (SOC) and carbon dioxide (CO₃) fluxes in agricultural ecosystems. However, microbial processes related to SOM decomposition have not been, or are inadequately, represented in these models, limiting predictions of SOC responses to changes in microbial activities. In this study, we developed a microbial-mediated decomposition model based on a widely used biogeochemical model, DeNitrification-DeComposition (DNDC), to simulate C dynamics in agricultural ecosystems. The model simulates organic matter decomposition, soil respiration, and SOC formation by simulating microbial and enzyme dynamics and their controls on decomposition, and considering impacts of climate, soil, crop, and farming management practices (FMPs) on C dynamics. When evaluated against field observations of net ecosystem CO₂ exchange (NEE) and SOC change in two winter wheat systems, the model successfully captured both NEE and SOC changes under different FMPs. Inclusion of microbial processes improved the model's performance in simulating peak CO₂ fluxes induced by residue return, primarily by capturing priming effects of residue inputs. We also investigated impacts of microbial physiology, SOM, and FMPs on soil C dynamics. Our results demonstrated that residue or manure input drove microbial activity and predominantly regulated the CO, fluxes, and manure amendment largely regulated long-term SOC change. The microbial physiology had considerable impacts on the microbial activities and soil C dynamics, emphasizing the necessity of considering microbial physiology and activities when assessing soil C dynamics in agricultural ecosystems.

Plain Language Summary Soil microbes drive decomposition of soil organic matter (SOM) and regulate soil carbon (C) dynamics. Process-based models are useful tools for quantifying changes in soil organic carbon (SOC) and carbon dioxide (CO₂) fluxes in agricultural ecosystems. However, microbial processes related to SOM decomposition have not been, or are inadequately, represented in these models, limiting predictions of SOC responses to changes in microbial activities. We developed a microbial-mediated decomposition model based on a widely used biogeochemical model, DeNitrification-DeComposition (DNDC), to simulate C dynamics in agricultural ecosystems. The model simulates organic matter decomposition, soil respiration, and SOC formation by simulating microbial dynamics and controls on decomposition, and considering impacts of climate, soil, crop, and farming management practices (FMPs) on C dynamics. We also investigated impacts of microbial physiology, SOM, and FMPs on soil C dynamics. Our results demonstrated that residue or manure input drove microbial activity and predominantly regulated CO, fluxes, and manure amendment largely regulated long-term SOC change. The microbial physiology had considerable impacts on microbial activities and soil C dynamics, emphasizing the necessity of considering microbial physiology and activities when assessing soil C dynamics in agricultural ecosystems. These results provide insights in simulating microbial-mediated soil C dynamics in agricultural ecosystems.



1. Introduction

Soils contain more carbon (C) than plant biomass and the atmosphere combined at the global scale (IPCC, 2013; Lehmann & Kleber, 2015). Through soil C sequestration or converting and releasing soil or ganic carbon (SOC) into atmospheric carbon dioxide (CO_2) or methane (CH_4), soils exert large impacts on global climate change (IPCC, 2013; Lal, 2004). Moreover, soil organic matter (SOM) contains nutrients that are essential for plant growth. Therefore, SOM dynamics influence the availability of soil nutrients, thereby impacting plant productivity in terrestrial ecosystems (Lal, 2008). Despite considerable research, large uncertainty remains over the spatiotemporal variability and magnitude of the SOC changes and the associated CO_2 fluxes in agricultural ecosystems due to complex processes and interactions involved (e.g., Dignac et al., 2017; Lal, 2018). At the same time, there is a considerable interest in promoting soil C sequestration in agricultural soils (e.g., Amelung et al., 2020; Smith, 2004), which will require advances in methods for measurement and monitoring, increased process-level understanding, improved process-based modeling of agricultural soils and impacts of farming activities, and expanded and enhanced geospatial databases of soil properties and farming management (e.g., Paustian et al., 2016).

Changes of SOC in agricultural ecosystems ultimately depend on the balance between C inputs (e.g., plant litter, organic manure) and outputs (e.g., SOC losses due to decomposition or erosion), which are affected by farming management practices (FMPs), plant biomass production, and SOM formation and decomposition. Agricultural management practices, such as fertilization, input of crop straw, and organic manure amendment, have been regarded as important regulators of the SOC changes, although the trend and magnitude of the SOC changes in response to agricultural activities are highly variable (e.g., Liu et al., 2014; Maillard & Angers, 2014; Powlson et al., 2012). In addition, SOC changes and CO_2 fluxes from decomposition are largely controlled by soil microbial physiology and activity, given that soil microbes play a central role in SOM decomposition, formation, and depletion (Kallenbach et al., 2016; Miltner et al., 2012). However, the responses of microbial-mediated decomposition processes to agricultural activities and the impacts of microbial physiology on SOC changes remain unclear (Kallenbach et al., 2016; Prommer et al., 2020).

Process-based biogeochemical SOC models such as DeNitrification-DeComposition (DNDC), DayCent, and Roth-C have been developed and applied to quantify SOC dynamics and CO₂ fluxes in agricultural ecosystems as well as evaluate their responses to variations in FMPs and environmental factors (e.g., Parton et al., 2015; Smith et al., 1997; Van Wesemael et al., 2010). While many improvements have been made in modeling agricultural SOC dynamics by incorporating farming activities, vegetation, and biogeochemical processes related to C cycling into the framework of biogeochemical models (e.g., Li et al., 1994), key limitations still exist. For example, in most biogeochemical models SOC decomposition is simulated as a first-order process and is directly proportional to the size of SOC pool. Microbial processes important to SOC decomposition and their associated microbial communities often have not been, or are inadequately, represented in these models, which limit the model's capacity of predicting SOC responses to changes in microbial activities (Georgiou et al., 2017; Sulman et al., 2014). In recent years, several microbial models of SOC decomposition have been proposed or developed to predict soil C dynamics by explicitly representing microbial or enzymatic decomposition of SOC (e.g., Abramoff et al., 2017; Allison et al., 2010; Gao et al., 2020; Guo et al., 2020; Wang et al., 2013; Wieder et al., 2015). Microbial SOC models have incorporated processes to simulate microbial dynamics and feedbacks between SOC changes and microbial activities. However, they usually lack detailed processes to explicitly simulate plant growth and litter production, and have not incorporated FMPs (e.g., harvest, tillage, fertilization, irrigation, organic manure amendment) that influence soil C dynamics (e.g., Sulman et al., 2018), limiting their applications in agricultural ecosystems. Furthermore, there are large uncertainties regarding their behaviors in modeling SOC changes and CO₂ fluxes due to very limited model testing against field-scale observations.

The process-based biogeochemical model, DNDC, has incorporated a relatively complete suite of biophysical and biogeochemical processes and FMPs, which enables it to simulate vegetation growth, transport and transformations of C and nitrogen (N), SOC changes, and greenhouse gas (GHG) fluxes in terrestrial ecosystems (e.g., Deng et al., 2018, 2020; Gilhespy et al., 2014; Giltrap et al., 2010; Li, 2000; Li et al., 1992a, 2012). However, DNDC does not explicitly simulate microbial or enzymatic decomposition of SOM, and therefore is unable to reproduce microbial regulations on soil C cycle. In this study, our objectives were: (a) to develop a microbe-driven SOM decomposition model, based on the DNDC model, to predict microbial activities



and C dynamics, and evaluate the model's performance in quantifying CO_2 fluxes and SOC changes from agricultural ecosystems and (b) to explore the impacts and evaluate the relative importance of FMPs, SOM, and microbial physiology on soil microbial activity, SOC change, and CO_2 flux.

2. Materials and Methods

2.1. The DNDC Model

DNDC is a process-based biogeochemical model developed for quantifying ecosystem C sequestration and the exchange of C and N gases between terrestrial ecosystems and the atmosphere (Li, 2000; Li et al., 1992a, 1992b). The model has been extensively evaluated against datasets of SOC changes and GHG fluxes measured worldwide (Gilhespy et al., 2014; Giltrap et al., 2010). DNDC is comprised of six interacting sub-models: soil climate, plant growth, decomposition, nitrification, denitrification, and methanogenesis. The soil climate, plant growth, and decomposition sub-models convert the primary drivers, such as climate, soil properties, vegetation, and anthropogenic activity, into soil environmental factors, such as soil temperature and moisture, pH, redox potential, and concentration of substrates for the simulated biogeochemical processes. The nitrification, denitrification, and methanogenesis sub-models simulate C and N transformations that are mediated by soil microbes and controlled by soil environmental factors and substrate concentrations (Li, 2000; Li et al., 2012).

In DNDC, net ecosystem CO₂ exchange (NEE) is calculated as the difference between net primary production (NPP) and soil microbial heterotrophic respiration. NPP is simulated at a daily time step by considering impacts of several environmental factors (e.g., solar radiation, air temperature, soil moisture, and N availability) on plant growth (Deng et al., 2014). The impacts of solar radiation, soil moisture, or N availability on NPP are calculated based on the potential photosynthetically active radiation (PAR), water, or N required by optimum plant growth and the availability of these factors. The model simulates the production of plant biomass and litter, and incorporates the residue litter into the SOM pools. Soil heterotrophic respiration is simulated by tracking the decomposition of SOM. The model divides SOM into four organic pools: litter, living microbes, humads (i.e., active humus), and passive humus. Each pool is further divided into labile and resistant sub-pools with specific C:N ratios and decomposition rates (Figure 1). In DNDC, the size of the living microbes pool does not affect the decomposition rates of other SOM pools. Decomposition of each SOM pool depends on its specific decomposition rate and pool size, as well as soil thermal, moisture, and mineral N conditions (Li et al., 1994, 2012). When organic matter, such as litter or organic manure, has been applied into the soil, the model partitions it into different litter and/or humads pools based on its C:N ratio (Figure 1; Li et al., 1994, 2012). Organic C is converted to CO₂, microbial biomass, humads, and finally passive humus through decomposition. Labile C is gradually lost and resistant C becomes relatively more abundant in the soil during decomposition of exogenous organic matter (Figure 1). The model also simulates soil N dynamics by tracking a series of biogeochemical reactions: decomposition (mineralization), microbial immobilization, plant uptake, ammonia volatilization, ammonium adsorption, nitrification, denitrification, nitrate leaching, and fluxes of N gases (i.e., NH₂, NO, N₂O, and N₂). Soil mineral N concentration is calculated at a daily time step and is a factor regulating SOM decomposition (Li et al., 1992a). Further details regarding the DNDC structure, inputs, and outputs, as well as the physical, chemical, and biogeochemical processes incorporated into the model's framework, are available in Gilhespy et al. (2014), Li (2000), and Li et al. (2012).

2.2. Modification of DNDC

In this study, we improved the DNDC model by explicitly simulating the microbe-driven SOM decomposition. The improved model simulates cycling and storage of multiple SOM pools by explicitly simulating microbial and enzyme dynamics and their controls on SOM decomposition, in addition to simulating impacts on C dynamics of crop growth and common FMPs in agricultural ecosystems (Figure 1).

2.2.1. SOM Pools and Decomposition

The model primarily simulates decomposition of six SOM pools (Figure 1). The very labile, labile, and resistant litter pools are organic matter from plant residue or organic manure amendments, and have different





Figure 1. Structure of the soil organic matter (SOM) decomposition module in the modified DNDC model. Boxes represent simulated organic matter pools, solid arrows represent fluxes among the pools, and P1 to P10 represent simulated processes. The model simulates decomposition of six SOM pools (green boxes). SOM pool decomposition rates are directly controlled by the soil enzyme concentration, which is regulated by dynamics of soil living microbes, dissolved organic carbon, and dissolved organic nitrogen (collectively DOM in the figure).

C:N ratios and relatively high, moderate, and low potential decomposition rates, respectively. The other three SOM pools are labile humads, resistant humads, and passive humus. The humads pools receive products from SOM decomposition or microbial death, and have different potential decomposition rates as well. The model simulates decomposition of these six pools, with the decomposition rate of each pool directly regulated by soil enzyme concentration (Equation 1, below, and Equations A1–A7 in Appendix A). During decomposition, decomposed litter and SOM are primarily converted into dissolved organic carbon (DOC) and dissolved organic nitrogen (DON). The DOC and DON can be assimilated by soil microbes to form microbial biomass, released as CO_2 and mineralized into ammonium (NH₄-N), respectively, and/or leached out of the soil. The model also explicitly simulates dynamics of soil microbial and enzyme pools, and dead microbes and decayed enzyme are major components of humads and an important source of DOC and DON (Figure 1). In addition, the model considers C flows from humads to humus to represent humification of the humads pools (Li et al., 1992a; Molina et al., 1983).

The basic concept underlying a microbial-driven decomposition model is that the decomposition of SOM is controlled by the activity of exoenzymes, instead of simply concentration of SOM (Schimel & Weintraub, 2003). Specifically, the rate of decomposition of each SOM pool is simulated using the Arrhenius and Michaelis-Menten equations (e.g., Allison et al., 2010; Wang et al., 2013); with the rate directly controlled by concentrations of soil enzyme and each SOM pool, soil temperature and moisture, and clay content (Equation 1).

$$D_{\text{SOM}} = V \max_{\text{SOM}} \times \text{Enzyme} \times \frac{\text{SOM}}{K_{\text{SOM}} + \text{SOM}} \times f(T) \times f(W) \times f(\text{ST})$$
(1)

where D_{SOM} is the decomposition rate (mg C h⁻¹) of each SOM pool, $V \max_{\text{SOM}}$ is the maximum rate of SOM decomposition at reference temperature (20°C) when SOM concentration is not limiting, Enzyme is the aggregate concentration of soil exoenzymes (mg C g⁻¹) that is simulated based on soil microbial activities (Section 2.2.3), SOM is the concentration of each SOM pool (mg C g⁻¹), K_{SOM} is the Michaelis half-saturation constant for decomposition of each SOM pool, and f(W) and f(ST) are functions calculating the impacts



Table 1

Model Parameters for Simulating Microbe and Enzyme Dynamics, and SOM Decomposition

Parameter	Description	Values	Sources
Parameters selecte	ed for sensitivity analysis		
Vmax _{DOCuptake}	Maximum uptake rate of DOC at a reference temperature of 20°C, mg C (mg microbe C) ⁻¹ hr^{-1}	0.24	Huang et al. (2018)
R _{MicrobeMaintenance}	Rate of microbial maintenance respiration, h ⁻¹	0.002	Calibrated ^a
$R_{ m MicrobeDeath}$	Microbial turnover rate, h ⁻¹	0.002	Allison et al. (2010)
CAE	Potential microbial carbon assimilation efficiency	0.6	Sinsabaugh et al. (2013)
$F_{\rm MICtoDOC}$	Fraction of dead microbial biomass that allocated to DOC	0.5	Allison et al. (2010)
			Wang et al. (2013)
R _{EnzymeProduction}	Enzyme production rate, h ⁻¹	10 ⁻⁵	Calibrated ^a
R _{EnzymeDecay}	Enzyme decay rate, h ⁻¹	10-3	Abramoff et al. (2017)
Vmax _{Litter_vl}	Maximum decomposition rate of very labile litter at reference temperature, mg C (mg Enzyme C)^{-1} $$\rm hr^{-1}$$	81	Wang et al. (2012)
$V \max_{\text{Litter}_l}$	Maximum decomposition rate of labile litter at reference temperature, mg C (mg Enzyme C)^{-1} hr^{-1}	81	Wang et al. (2012)
Vmax _{Litter_r}	Maximum decomposition rate of resistant litter at reference temperature, mg C (mg Enzyme C) ^{-1} hr ^{-1}	5.6	Calibrated ^b
$V \max_{Humads_l}$	Maximum decomposition rate of labile humads at reference temperature, mg C (mg Enzyme C) ^{-1} hr ^{-1}	13	Calibrated ^b
Vmax _{Humads_} r	Maximum decomposition rate of resistant humads at reference temperature, mg C (mg Enzyme C)^{-1} $$\rm hr^{-1}$$	1.3	Calibrated ^b
Vmax _{Humus}	Maximum decomposition rate of humus at reference temperature, mg C (mg Enzyme C) ⁻¹ hr ⁻¹	0.43	Calibrated ^b
Other parameters			
Ea_{Litter_vl}	Activation energy for decomposition of very labile litter in the Arrhenius equation, KJ mol ⁻¹	37	Wang et al. (2012, 2013)
Ea_{Litter_l}	Activation energy for decomposition of labile litter in the Arrhenius equation, KJ mol $^{-1}$	37	Wang et al. (2012, 2013)
Ea_{Litter_r}	Activation energy for decomposition of resistant litter in the Arrhenius equation, KJ mol^{-1}	53	Wang et al. (2012, 2013)
Ea_{Humads_l}	Activation energy for decomposition of labile humads in the Arrhenius equation, KJ mol^{-1}	47	Wang et al. (2012, 2013)
Ea_{Humads_r}	Activation energy for decomposition of resistant humads in the Arrhenius equation, KJ mol $^{-1}$	47	Wang et al. (2012, 2013)
Ea _{Humus}	Activation energy for decomposition of humus in the Arrhenius equation, KJ mol $^{-1}$	53	Wang et al. (2012, 2013)
K_{Litter_vl}	Michaelis half-saturation constant for decomposition of very labile litter, mg C g ⁻¹ soil	1.20	Calibrated ^c
K_{Litter_l}	Michaelis half-saturation constant for decomposition of labile litter, mg C $\rm g^{-1}$ soil	1.20	Calibrated ^c
K_{Litter_r}	Michaelis half-saturation constant for decomposition of resistant litter, mg C g ⁻¹ soil	12.0	Calibrated ^c
K_{Humads_l}	Michaelis half-saturation constant for decomposition of labile humads, mg C g ⁻¹ soil	12.0	Calibrated ^c
K _{Humads_r}	Michaelis half-saturation constant for decomposition of resistant humads, mg C g $^{-1}$ soil	60.0	Calibrated ^c
K _{Humus}	Michaelis half-saturation constant for decomposition of humus, mg C $\rm g^{-1}$ soil	100.0	Calibrated ^c
K _{DOC}	Michaelis DOC half-saturation constant for DOC uptake, mg C g^{-1} soil	0.30	Calibrated ^d
K _{O2}	Michaelis O ₂ half-saturation constant for DOC uptake, mmol cm ⁻³	0.00015	Li (2016)

 ${}^{a}R_{\text{MicrobeMaintenance}}$ and $R_{\text{EnzymeProduction}}$ were calibrated within their uncertainty ranges of 10^{-4} to 8×10^{-3} hr⁻¹ and 10^{-5} to 8×10^{-5} hr⁻¹, respectively (He et al., 2015). ${}^{b}V\text{max}_{\text{Litter}_{T}}$ was calibrated within the uncertainty range of 0.2–33.0 mg C (mg Enzyme C)⁻¹ hr⁻¹, and $V\text{max}_{\text{Humads}_{T}}$, $V\text{max}_{\text{Humads}_{T}}$, and $V\text{max}_{\text{Humads}_{T}}$, and $V\text{max}_{\text{Humads}_{T}}$, and $V\text{max}_{\text{Humads}_{T}}$, and $V\text{max}_{\text{Humads}_{T}}$, $M\text{max}_{\text{Humads}_{T}}$, $M\text{max}_{\text{Humads}_{T}$, $M\text{max}_{\text{Humads}_{T}}$, $M\text{max}_{\text{Humads}_{T}$, $M\text{max}_{\text{Humads}_{T}}$, $M\text{max}_{\text{Humads}_{T}$, $M\text{max}_{\text{Humads}_$

on SOM decomposition of soil water content and clay content, respectively (Equations A9 and A10). The $V_{\text{Max}_{\text{SOM}}}$ values are different for each SOM pool (Table 1), and the maximum rate of SOM decomposition under a specific temperature is calculated using the Arrhenius equation (f(T); Equation A8).

2.2.2. DOC and DON Dynamics and Soil Microbial C and N Uptakes

Soil DOC and DON are produced through decomposition of each SOM pool, from death of microbes and decay of enzymes, added through root exudation, and/or through amendments of organic manure (e.g., slurry manure). They can be taken up and used by soil microbes for biomass production and respiration

(Schimel & Weintraub, 2003), or leached out of soils. In addition, the model simulates DOC consumption by denitrification (Li et al., 1992a) and methanogenesis (Deng et al., 2017) as well as DON mineralization. Specifically, DOC and DON dynamics are simulated as (Equations 2 and 3):

$$\frac{d\text{DOC}}{dt} = \text{DOC}_{\text{Manure}} + \text{DOC}_{\text{Root}} + D_{\text{Litter}} + D_{\text{Humads}} \times F_{\text{HumadsToDOC}} + D_{\text{Humus}} + \text{Death}_{\text{Microbe}} \times F_{\text{MICToDOC}} + D_{\text{Enzyme}} - \text{DOC}_{\text{Uptake}} - \text{DOC}_{N} - \text{DOC}_{M} - \text{DOC}_{\text{Leaching}}$$
(2)

$$\frac{d\text{DON}}{dt} = \text{DON}_{\text{Manure}} + \text{DON}_{\text{Root}} + \frac{D_{\text{Litter}}}{\text{CN}_{\text{Litter}}} + \frac{D_{\text{Humads}} \times F_{\text{HumadsToDOC}}}{\text{CN}_{\text{Humads}}} + \frac{D_{\text{Humus}}}{\text{CN}_{\text{Humus}}} + \frac{D_{\text{enzyme}}}{\text{CN}_{\text{Humos}}} + \frac{D_{\text{CN}_{\text{Humus}}}}{\text{CN}_{\text{Humus}}} + \frac{D_{\text{Humus}}}{\text{CN}_{\text{Humus}}} + \frac{D_{\text{Hu$$

where $\text{DOC}_{\text{Manure}}$ and $\text{DON}_{\text{Manure}}$ are DOC and DON added through organic manure and are calculated based on model inputs of manure amendment rate, manure type, and C:N ratio of manure (Li, 2016; Li et al., 2012); DOC_{Root} is the DOC added through root exudation, which is simulated as 45% of the C transferred to roots from photosynthetic production (Zhang et al., 2002), DON_{Root} is the DON added through root exudation, which is calculated as the DOC_{Root} divided by the C:N ratio of root exudation. D_{Litter} , D_{Humads} , and D_{Humus} are decomposition of litters, humads, and humus, respectively (Equations A1–A7), Death_{Microbe} and D_{Enzyme} are the microbe death and enzyme decay at each time step, $\text{CN}_{\text{Litter}}$, $\text{CN}_{\text{Humads}}$, $\text{CN}_{\text{Microbe}}$, and $\text{CN}_{\text{Enzyme}}$ are the C:N ratios of litters, humads, humus, soil microbes, and enzyme respectively, $F_{\text{HumadsToDOC}}$ and F_{MICTODOC} are the fractions of decomposed humads and dead microbes that allocated to DOC, respectively, $\text{DOC}_{\text{Uptake}}$ and $\text{DON}_{\text{Uptake}}$ are the DOC and DON uptakes at each time step, DOC_N and DOC_M are the DOC consumed through denitrification (Li et al., 1992a) and methanogenesis (Deng et al., 2017), respectively, and $\text{DOC}_{\text{Leaching}}$ and $\text{DON}_{\text{Leaching}}$ are the DOC and DON leached out of soils, which are simulated based on subsurface drainage flows and DOC and DON concentrations (Deng et al., 2011; Li et al., 2006). Note that DOC and DON flux rates in Equations 2 and 3 are regulated by soil environmental factors and are zero under some conditions (e.g., DOC_M is zero in uplands).

DOC uptake by soil microbes is modeled based on the Arrhenius and Michaelis-Menten equations (Abramoff et al., 2017; Allison et al., 2010) with the potential uptake rate ($DOC_{PUptake}$) controlled by the soil temperature, soil living microbes, and soil DOC and oxygen concentrations (Equation 4).

$$DOC_{PUptake} = V \max_{DOCUptake} \times Microbe \times \frac{DOC}{K_{DOC} + DOC} \times \frac{O_2}{K_{O_2} + O_2} \times f(T)$$
(4)

where Vmax_{DOCUptake} is the maximum rate of DOC uptake at reference temperature (20°C) when DOC and oxygen concentration are not limiting, DOC and O₂ are the soil DOC and oxygen concentrations, respectively, and K_{DOC} and K_{O2} are their corresponding Michaelis half-saturation constants. The maximum rate of DOC uptake under a specific temperature is calculated using the Arrhenius equation (f(T); Equation A8).

The actual DOC uptake is constrained by soil DOC and DON availability, and is calculated as (Equation 5):

$$DOC_{Uptake} = DOC_{PUptake}$$
 If No DOC and DON limitation
$$DOC_{Uptake} = DOC$$
 If limited by soil DOC or DON availability (5)

After determining the $\text{DOC}_{\text{Uptake}}$, the actual DON uptake is calculated as $\text{DOC}_{\text{Uptake}}/\text{CN}_{\text{DOC}}$, where CN_{DOC} is the C:N ratio of DOC. Please note that the actual DON uptake is constrained by soil DON availability because the CN_{DOC} is calculated using the DOC and DON concentrations at each time step.

2.2.3. Dynamics of Microbes and Enzyme

The model explicitly simulates soil microbial dynamics by considering growth and mortality of microbes, and production of enzymes (Equation 6):



....

$$\frac{d\text{Microbe}}{dt} = \text{Growth}_{\text{Microbe}} - \text{Death}_{\text{Microbe}} - P_{\text{Enzyme}}$$
(6)

where $\text{Growth}_{\text{Microbe}}$ is the growth of microbial biomass, $\text{Death}_{\text{Microbe}}$ is the microbe mortality, P_{Enzyme} is the enzyme production.

Microbial growth is controlled by DOC uptake, microbial maintenance respiration, microbial carbon assimilation efficiency (CAE; defined as the maximum fraction of DOC uptake that can be allocated to microbial biomass), and availability of total dissolved nitrogen (DN) in soils (Schimel & Weintraub, 2003). If DOC uptake exceeds maintenance respiration, the excess C can be assimilated by soil microbes. After meeting the demand of microbial growth, the element in excess is released, by growth or overflow respiration for C or by N mineralization (Manzoni et al., 2012; Schimel & Weintraub, 2003). The actual growth of microbial biomass is also constrained by the DN availability. Specifically, the model simulates the microbial growth as:

$$Growth_{Microbe} = \left(DOC_{Uptake} - M_{Maintenance}\right) \times CAE \qquad If \frac{DN}{N_{Demand}} \ge 1.0$$

$$Growth_{Microbe} = \left(DOC_{Uptake} - M_{Maintenance}\right) \times CAE \times \frac{DN}{N_{Demand}} \qquad If \frac{DN}{N_{Demand}} < 1.0$$
(7)

where $M_{\text{Maintenance}}$ is the microbial maintenance respiration, *DN* is total dissolved nitrogen, including both DON and mineral N from either SOM mineralization or inputs through FMPs (e.g., fertilization), is simulated through the N biogeochemical reactions in DNDC, and N_{Demand} is the potential N required for microbe growth.

The microbial maintenance respiration is simulated as a constant fraction ($R_{\text{MicrobeMaintenance}}$) of living microbial biomass (Equation 8):

$$M_{\text{Maintenance}} = R_{\text{Microbe Maintenance}} \times \text{Microbe}$$
(8)

The potential N required for microbial growth is determined by potential microbial growth without DN limitation and the microbial C:N ratio (Equation 9):

$$N_{\text{Demand}} = \left(\text{DOC}_{\text{Uptake}} - M_{\text{Maintenance}}\right) \times \text{CAE} / \text{CN}_{\text{Microbe}}$$
(9)

The mortality rate of soil microbes is modeled as a density-dependent process of living biomass (Abramoff et al., 2017) (Equation 10):

$$Death_{Microbe} = R_{MicrobeDeath} \times Microbe^2$$
(10)

The enzyme concentration directly controls the decomposition of each SOM pool, and is determined by the enzyme production (P_{Enzyme}) and decay (D_{Enzyme}) at each time step (Equation 11).

$$\frac{d\text{Enzyme}}{dt} = P_{\text{Enzyme}} - D_{\text{Enzyme}}$$
(11)

By following Allison et al. (2010), we modeled enzyme production as a constant fraction ($R_{\text{EnzymeProduction}}$) of living microbial biomass (Equation 12) and modeled enzyme decay as a first-order process with a rate constant $R_{\text{EnzymeDecay}}$ (Equation 13).

$$P_{\text{Enzyme}} = R_{\text{EnzymeProduction}} \times \text{Microbe}$$
(12)

$$D_{\text{Enzyme}} = R_{\text{EnzymeDecay}} \times \text{Enzyme}$$
(13)



Table 2

Primary Model Input Parameters for the Rothamsted and Oklahoma Winter Wheat Sites

Model parameters ^a	Rothamsted	Oklahoma
Soil bulk density, g cm ⁻³	1.25	1.3
Clay content, %	25	21
Initial SOC content ^d , %	1.0	1.6
pH	7.7	5.9
Fraction of crop residue return ^d	0.5	1.0
N fertilization ^{b, d} , kg N ha ⁻¹	0 or 144	72 or 62
Organic manure application ^{c, d} , kg C ha ⁻¹	0 or 2,800	0
C:N ratio of manure ^d	12.5	None

^aModel input parameters were from field records. ^bAmount of synthetic N fertilizer applied into fields was 0 kg N ha⁻¹ under the control and FYM treatments and 144 kg N ha⁻¹ under the NPK treatment at the Rothamsted site, and was 72 kg N ha⁻¹ in the 2014 to 2015 crop season and 62 kg N ha⁻¹ in the 2015 to 2016 crop season at the Oklahoma site. ^cAmount of organic manure applied on Rothamsted fields was 0 kg C ha⁻¹ under the FYM treatment. ^dThe parameters selected for sensitivity analysis.

2.3. Field Data

2.3.1. SOC Data From the Rothamsted Experiment

To evaluate the performance of the modified DNDC in simulating longterm SOC changes, we used SOC observations from the Rothamsted classical broadbalk winter wheat experiment (Powlson et al., 2012). The experiment started in 1843, and field observations at Harpenden in Hertfordshire in Southern England (N51°48′36″, W0°22′30″) included SOC changes of winter wheat croplands during 1843–2010. The soil is classified as Chromic Luvisol in the FAO Soil Classification, or Aquic Paleudalf in the U.S. Soil Taxonomy. The soil (0–23 cm depth) has a pH of 7.0–7.5, and its texture is 25% sand, 50% silt, and 25% clay (www.era.rothamsted. ac.uk/Broadbalk/bbksoils).

Winter wheat has been grown continuously since 1843, except for occasional fallows during the period from 1926 to 1966 to control weeds. Wheat was sown in autumn and harvested in summer of the next year. We evaluated the model using field data of SOC changes under three treatments: control, conventional fertilizer application (NPK), and farmyard manure amendments (FYM). The control treatment has received no fertilizer or organic manure since 1843. The NPK is local conventional practice of applying 144 kg N ha⁻¹ yr⁻¹, 35 kg P ha⁻¹ yr⁻¹, and 90 kg K ha⁻¹ yr⁻¹. The FYM treatment has received farmyard manure at a rate of 35 metric ton ha⁻¹ yr⁻¹ (about 2,800 kg C ha⁻¹ yr⁻¹) since 1843 (Table 2). Changes in SOC (0–23 cm), climate, soil properties, and FMPs were recorded during the experimental period. Long-term SOC changes under the three treatments indicated that both application of synthetic

fertilizers and amendment of organic manure caused increasing SOC. After about 150 years, the SOC stock (0-23 cm) at the NPK plot was about 0.25 times (25%) higher than the control plot, and the SOC stock (0-23 cm) at the FYM plot was about 2.5 times higher than the control plot (Johnston et al., 2009). The long-term SOC data provided a unique data set for assessing the impacts of fertilization and manure management on SOC changes and evaluating the model's capability in predicting the SOC responses to these different FMPs.

2.3.2. NEE Data From Winter Wheat Eddy Flux Tower Sites in Oklahoma

Field data used to evaluate the simulations of NEE from the modified DNDC model were measured from a winter wheat cropland at the US Department of Agriculture-Agricultural Research Service (USDA-ARS), Grazinglands Research Laboratory (GRL, N35°34'7, W98°3'21"), in El Reno, Oklahoma. This area has a temperate continental climate with an average air temperature of 14.9°C and an average annual rainfall of 860 mm from 1971 to 2000 (Fischer et al., 2012). Field data included daily NEE measured during October 2014 to September 2016 (Bajgain et al., 2018). The soil of the study field is characterized as deep, well-drained, loam with clayey or loamy subsoil. During the study period, wheat was sown in September and was terminated in June of the next year, and the field was kept fallow during the summer seasons with weed control by tillage and herbicide application. The field received C inputs through wheat residue return and N inputs through synthetic N fertilizations during the study period (Table 2). Continuous CO_2 fluxes were measured using an eddy covariance (EC) tower, and daily and annual values of NEE (positive values represent net CO_2 fluxes into the atmosphere and negative fluxes represent net CO_2 uptake by the wheat field) were used for evaluating the model. In addition to NEE measurements, climate, soil properties, and FMPs for the wheat field were also recorded.



2.4. Model Application

2.4.1. Model Evaluation

The modified DNDC was run for the Rothamsted and Oklahoma winter wheat sites to evaluate the model's performance in simulating long-term SOC changes and seasonal to annual CO_2 fluxes, respectively. The Rothamsted site was simulated from 1840 to 2010, and the Oklahoma site was simulated from 2014 to 2016. Daily meteorological data required for driving the model (i.e., air temperature and precipitation for Rothamsted and air temperature, precipitation, and solar radiation for Oklahoma) were obtained from on-site measurements for both sites. The primary soil input parameters, including soil texture, clay fraction, bulk density, pH, initial SOC content, were also determined using on-site records (Table 2). The input parameters of FMPs, including planting and harvest dates, tillage, fertilization, wheat residue return, and amendments of organic manure, were estimated from field records (Table 2).

New model parameters are needed to simulate microbial and enzyme dynamics as well as decomposition of each SOM pool. These parameters were set either to literature values or calibrated against the field observations of SOC changes under NPK in the Rothamsted classical broadbalk winter wheat experiment through the trial and error method (Table 1). The calibrated parameters included rates of microbial maintenance respiration ($R_{\text{MicrobeMaintenance}}$) and enzyme production ($R_{\text{EnzymeProduction}}$), maximum decomposition rates of resistant litter ($V \max_{\text{Litter_r}}$), labile humads ($V \max_{\text{Humads_l}}$), resistant humads ($V \max_{\text{Humads_r}}$), and humus ($V \max_{\text{Humads}}$) at reference temperature, and Michaelis half-saturation constants for decomposition of each SOM pool and DOC uptake.

 $R_{\text{MicrobeMaintenance}}$ and $R_{\text{EnzymeProduction}}$ were estimated by calibrating simulated SOC changes with observed SOC changes under the NPK treatment at the Rothamsted site. The maximum decomposition rates at 20°C were estimated as 5.6, 13, 1.3, and 0.43 g C (g Enzyme C)⁻¹ hr⁻¹ for the resistant litter, labile humads, resistant humads, and humus pools, respectively, by calibrating against observed SOC changes under NPK. The values of these parameters were within their reported uncertainties (Wang et al., 2013) and generally reflected the trend of decreased potential decomposition rate from labile humads to resistant litter and humads, and to humus in DNDC (Li et al., 1992a).

The Michaelis half-saturation constants for decomposition of each SOM pool were estimated after fixing their corresponding maximum decomposition rates. We calibrated the Michaelis half-saturation constants (Table 1) by considering the reported uncertainties of these parameters (Huang et al., 2018; Wang et al., 2013) and the steady-state concentrations of each SOM pool by referring to the analysis by Wang et al. (2013). In general, the input parameters described above were primarily determined by calibrating against observed SOC changes under NPK at the Rothamsted site, and the calibrated model was then validated by comparing simulations against observations of SOC changes under the control and FYM treatments at Rothamsted, and NEE at the Oklahoma site. The new model parameters for simulating microbe and enzyme dynamics, and SOM decomposition (Table 1) were identical between the Rothamsted and Oklahoma sites.

Two statistical measures, the relative root mean squared error (RRMSE) and the correlation coefficient (R), were used to quantify the accordance and correlation between model predictions and field observations (Moriasi et al., 2007).

2.4.2. Impacts of FMPs, SOM, and Microbial Physiology on Soil Carbon Dynamics

To investigate the impacts of FMPs, SOM, and microbial physiology on soil microbial activity, CO_2 fluxes, and SOC changes, we conducted a series of simulations with the modified DNDC by varying 18 parameters (Tables 1 and 2). These parameters are relevant to FMPs, initial SOC content, maximum SOM decomposition rate, or microbial physiology, and can impact SOC change and CO_2 flux from soil heterotrophic respiration (e.g., Six et al., 2006). The baseline was set by referring to the environmental conditions and FMPs under the control treatment of the Rothamsted classical broadbalk winter wheat experiment. We conducted simulations for the years from 1840 to 2010 (around 170 years), with continuous cultivation of winter wheat during these years, which was slightly different with the setting of occasional fallow years under the control treatment. In the baseline simulation, there was no fertilization and manure amendment, and the fraction of returned crop residue was set to 0.5. In order to fully represent potential combinations of the selected parameters, we created alternative scenarios by varying these parameters simultaneously.





Figure 2. Simulated (lines) and observed (circles) SOC (0–23 cm) from 1840 to 2010 under the control, fertilizer (NPK), and farmyard manure (FYM) treatments at the winter wheat field in the Rothamsted Agricultural Station, Harpenden, UK.



Figure 3. Simulated (lines) and observed (circles) daily net ecosystem exchange (NEE) of CO_2 during (a) October 2014 through September 2015 and (b) October 2015 through September 2016, at a winter wheat field in El Reno, OK, USA. The black arrows indicate the dates of roots and residues return and the blue arrows indicate the dates of tillage events. The correlations between the simulated and observed daily NEE were significant for all cases (P < 0.001). The gray lines are simulations of daily NEE from the original DNDC model that does not explicitly simulate microbial decomposition of soil organic matter. The R values between the original DNDC's simulations and observations of daily NEE were 0.70 and 0.77 in the rotational years from 2014 to 2015 and 2015 to 2016, respectively.

The variations for the parameters related to FMPs were set generally by referring to the farming management in the Rothamsted experiment, and the ranges were from 0.0 to 1.0 for the fraction of the returned crop residue, 0-144 kg N ha⁻¹ year⁻¹ for N fertilizer application rate, 0-2,800 kg C ha⁻¹ year⁻¹ for organic manure input, and 10 to 15 for manure C/N ratio. For the other parameters, we set two ranges of variation ($\pm 20\%$ and $\pm 50\%$ of the baseline value) to represent relatively small and large variations for parameters related to SOM or microbial physiology. These two sets of scenarios were denoted as V20% and V50%, respectively. The changes of all these parameters were randomly picked from their corresponding ranges, assuming uniform frequency distributions. Two thousand scenarios were run for each range of variations by using the Latin Hypercube Sampling strategy (Helton & Davis, 2003). The modified DNDC model was run from 1840 to 2010 with the varied parameters for both V20% and V50% (in total 4,000 runs). Simulated mean annual SOC changes, CO, fluxes from soil heterotrophic respiration, soil living microbe biomass, and enzyme pool size from 1840 to 2010 under different scenarios were analyzed.

In addition, we evaluated relative importance of the selected parameters to SOC change, CO_2 flux from soil heterotrophic respiration, and soil microbial activity by using a global sensitivity analysis, the Smirnov test (Saltelli et al., 2008; Tarantola & Becker, 2016). The Smirnov test can calculate a sensitivity index that quantifies the relative importance of the selected parameters (Tables 1 and 2) to outputs (i.e., SOC change, CO_2 flux from soil heterotrophic respiration, soil living microbes, and enzyme pool in this study) by fully exploring the variance space of the parameters. A parameter with higher sensitivity index is more sensitive (or important) than another parameter with lower sensitive index under their corresponding variations (Saltelli et al., 2008).

3. Results

3.1. Model Evaluation

3.1.1. SOC Change

At the Rothamsted site, changes of SOC storage were clearly different among the three treatments with different FMPs (Figure 2). During 1840 to 2010, the observed SOC storage decreased by about 15% under control, slightly increased by about 4% under NPK, and substantially increased by 172% under FYM. The amendment of FYM was identified as an important factor regulating the SOC changes through the Rothamsted experiment (Johnston et al., 2009; Powlson et al., 2012). Current SOC storage under the FYM treatment is substantially higher than under control and NPK, by 220% (78.4 vs. 24.5 Mg C ha^{-1}) and 162% (78.4 vs. 30.0 Mg C ha⁻¹), respectively. SOC storage was also influenced by the N fertilization (Figure 2), which was attributed to the higher organic matter returns through wheat residue in the NPK plot (Johnston et al., 2009). Simulations of SOC change were similar to corresponding field observations for both the NPK treatment used for model calibration and control and FYM treatments used for model validation (Figure 2), with the model capturing the differences of SOC change among the different treatments (Figure 2). From 1840 to 2010, simulated SOC storage decreased by 20% under control, increased by about 9% under NPK, and increased by 161% under FYM. These simulated SOC changes were in agreement with ob-





Figure 4. Ranges of simulated (a) average SOC change, (b) average annual soil heterotrophic respiration, (c) average annual mean living microbe mass, and (d) average annual mean enzyme pool size over a 170-year period under variations in selected parameters representing farming management practices, soil organic matter (SOM) properties, and microbe physiology. Parameters are defined in Tables 1 and 2. For parameters related to SOM or microbial physiology, variation ranges are within $\pm 20\%$ (V20%), or within $\pm 50\%$ (V50%). Bars show min and max values, asterisks indicate the 10th and 90th percentiles, boxes represent the bounds of 25th, 50th (median), and 75th percentiles, squares represent the average values, and blue triangles represent the baseline simulations.

servations for both model calibration and validation treatments. The RRMSE values were 7% for NPK (calibration), and 8% and 5%, respectively, for control and FYM (validation).

3.1.2. NEE

Simulated and observed daily NEE showed similar seasonal patterns at the OK winter wheat site, with net CO_2 uptake increasing following winter wheat planting, strong net CO_2 uptake for most days during wheat growing seasons from January to May, and net CO_2 release following wheat harvest (Figure 3). During summer fallow seasons, net CO_2 release occurred on most days, and peaks of CO_2 release were simulated and observed following the events of wheat root or residue incorporation and/or tillage. The simulated and observed daily NEE values were significantly correlated in both rotational years from 2014 to 2015 and 2015 to 2016 (P < 0.0001), with R values of 0.78 and 0.82 in the first and second rotational years, respectively. These results suggest that the modified DNDC model captured the seasonal patterns of daily NEE at the Oklahoma wheat field. The model also captured the magnitudes of the observed NEE. Annual total simulated NEE values were 495 and 389 kg CO_2 -C ha⁻¹, respectively, in the first and second rotational years, which were close to the corresponding observations of 559 and 397 kg CO_2 -C ha⁻¹. The RRMSE values were 13% in the first rotational year and 2% in the second rotational year.



Table 3

Correlation Coefficients Between Model Outputs, Including SOC Change, CO₂ Flux, Living Microbes, and Enzyme, and Selected Model Parameters That are Relevant to FMPs, SOM, or Microbial Physiology^a

	V20% ^b				V50% ^b			
	SOC change	CO ₂ flux	Living microbes	Enzyme	SOC change	CO ₂ flux	Living microbes	Enzyme
Vmax _{DOCuptake}	-0.052	-0.015	-0.009	0.007	-0.051	0.002	0.001	0.012
R _{MicrobeMaintenance}	-0.057	-0.016	-0.343*	-0.241*	-0.069	-0.012	-0.492*	-0.277*
$R_{ m MicrobeDeath}$	0.157*	0.017	-0.110*	-0.066*	0.217*	-0.014	-0.209*	-0.102*
CAE	0.245*	-0.078	0.424*	0.323*	0.365*	-0.167*	0.649*	0.390*
F _{MICtoDOC}	-0.242*	0.022	-0.004	0.001	-0.348*	0.083*	0.025	0.020
Vmax _{Litter_vl}	0.017	0.001	-0.004	-0.030	0.030	-0.001	-0.009	-0.033
Vmax _{Litter_l}	-0.001	-0.009	-0.013	0.016	-0.007	-0.012	-0.008	0.021
Vmax _{Litter_r}	0.039	0.021	0.018	0.025	0.037	0.014	0.012	0.020
$V \max_{Humads_l}$	0.017	0.032	0.011	0.021	-0.008	0.037	0.002	0.016
Vmax _{Humads_r}	0.014	0.020	0.004	0.016	0.010	0.023	0.004	0.011
Vmax _{Humus}	-0.252*	0.051	0.038	-0.006	-0.356*	0.112*	0.045	-0.018
SOC _{Initial}	-0.115*	0.050	0.056*	0.069*	-0.201*	0.075*	0.061*	0.067*
$F_{ m Residue}$	0.198*	0.571*	0.594*	0.443*	0.108*	0.549*	0.317*	0.199*
N fertilizer	0.023	-0.011	-0.012	-0.016	0.030	-0.010	-0.011	-0.016
Manure	0.758*	0.785*	0.486*	0.334*	0.425*	0.756*	0.283*	0.139*
CN _{Manure}	0.020	0.020	0.026	0.024	0.022	0.018	0.038	0.024
$R_{ m EnzymeProduction}$	-0.224*	0.046	-0.014	0.439*	-0.322*	0.097*	-0.024	0.517*
R _{EnzymeDecay}	0.244*	0.002	-0.032	-0.478*	0.345*	-0.050	-0.063*	-0.577*

^aThe "*" represents significant correlations (P < 0.01) between model outputs and selected parameters. ^bV20% and V50% denote variation range ±20% and ±50%, respectively, for parameters related to SOM or microbial physiology.

3.2. Impacts of Microbial Physiology, SOM, and FMPs on Carbon Dynamics

3.2.1. SOC Change

Figure 4a illustrates the simulated average annual SOC changes across the scenarios with different FMPs, SOM property, and microbial physiology. The SOC stock was predicted to be relatively stable under the baseline scenario. The simulated average annual SOC change rate was -42 kg C ha⁻¹ year⁻¹ (i.e., net C loss), and the simulated SOC stock was 14% lower than the initial stock after about 170 years under the baseline scenario (Figure 4a). The average annual change in simulated SOC varied from a net decrease of $-119 \text{ kg C} \text{ ha}^{-1} \text{ year}^{-1}$ to a net increase of 782 kg C ha⁻¹ year⁻¹ across the scenarios in V20% due to the variations in FMPs, SOM property, and microbial physiology (Figure 4a). Increasing the variations in initial SOM content, maximum SOM decomposition rate, and microbial physiology from $\pm 20\%$ to $\pm 50\%$ substantially increased the variations of the SOC change, with the simulated average SOC changes ranged between -264 kg C ha⁻¹ year⁻¹ (net decrease) and 1,582 kg C ha⁻¹ year⁻¹ (net increase) across the scenarios in V50% (Figure 4a). Simulated SOC stocks tend to be increasing along with the increases of wheat residue input, organic manure amendment, carbon assimilation efficiency (CAE), microbial turnover rate $(R_{\text{MicrobeDeath}})$, and enzyme decay rate $(R_{\text{EnzymeDecay}})$ (P < 0.01; Table 3), and the decreases of initial SOC stock, maximum decomposition rate of humus (Vmax_{Humus}), fraction of dead microbial biomass that allocated to DOC (F_{MICtoDOC}), and enzyme production rate ($R_{\text{EnzymeProduction}}$) (P < 0.01; Table 3). However, relatively strong correlations (|r| > 0.3) were calculated only for ±50% variations in CAE, $V_{\text{max}_{\text{Humus}}}$, F_{MICtoDOC} , $R_{\text{EnzymeProduction}}$, and $R_{\text{EnzymeDecay}}$ as well as amount of organic manure amendment.





Figure 5. Simulated average SOC change (kg C ha⁻¹ yr⁻¹; yellow boxes) and annual CO₂ flux (kg C ha⁻¹ yr⁻¹; blue boxes) from soil heterotrophic respiration across the 170 simulation years under selected scenarios with changes in fraction of wheat residue return and manure C input (different rows), and potential microbial carbon assimilation efficiency (CAE), and fraction of dead microbial biomass that allocated to dissolved organic carbon (F_{MICtoDOC}) (different columns). Values in the white boxes are litter (upper box) and manure (lower box) inputs (kg C ha⁻¹ yr⁻¹) for each case. Simulated change in SOC sequestration was primarily driven by C input, particularly by C input from manure amendments, and regulated by microbe physiology. Simulated SOC stocks tend to increase with increasing *CAE* and decreasing F_{MICtoDOC} . Note that litter inputs include C from root and are not equal under the scenarios with identical fraction (50% or 100%) of wheat residue return due to differences in wheat productivity.

3.2.2. CO₂ flux

The simulated average annual CO₂ flux from soil heterotrophic respiration was 1,919 kg CO₂-C ha⁻¹ yr⁻¹ across the roughly 170 simulation years in the baseline scenario. The simulations of average annual CO₂ flux varied from 1,497 to 5,707 kg CO₂-C ha⁻¹ yr⁻¹ across the scenarios in V20%. Increasing the variations in initial SOM stock, maximum SOM decomposition rate, and microbial physiology from ±20% to ±50% slightly increased the variation range of the average annual CO₂ flux by 8%–1,431–5,992 kg CO₂-C ha⁻¹ yr⁻¹ (Figure 4b), implying that the variation of annual CO₂ flux from soil heterotrophic respiration was primarily due to variations in FMPs. For example, increasing the organic manure input from 0 to 2,800 kg C ha⁻¹ yr⁻¹ and the fraction of wheat residue return from 0.5 to 1.0 nearly tripled the baseline CO₂ flux (5,730 vs. 1,919 kg CO₂-C ha⁻¹ yr⁻¹; Figure 5). Changing other parameters, such as CAE and F_{MICtoDOC} , resulted in smaller variations in the simulated average annual CO₂ fluxes were strongly correlated with the fraction of wheat residue return and organic manure incomparisons (r > 0.5, P < 0.01; Table 3).





Figure 6. Sensitivity of SOC change, soil heterotrophic respiration (CO₂ flux), living microbes, and enzyme pool to changes in selected parameters, with (left column) $\pm 20\%$ variation in initial soil organic matter (SOM) stock, maximum SOM decomposition rate, and microbial physiology parameters (V20%), and (right column) $\pm 50\%$ variation in initial SOM stock, maximum SOM decomposition rate, and microbial physiology parameters (V50%). Parameters are defined in Tables 1 and 2. The variable outcomes are more sensitive to parameters with higher sensitivity index values (i.e., they are more important) than to parameters with lower sensitive index values.

3.2.3. Microbial and Enzyme Dynamics

The simulated average annual mean living microbial and enzyme pools were 69.0 and 0.69 kg C ha⁻¹, respectively, across the 170 simulation years under the baseline scenario. The average annual mean living microbial and enzyme pools varied from 50.4 to 163.2 kg C ha⁻¹ and 0.41–2.07 kg C ha⁻¹, respectively, across all the scenarios in V20% (Figures 4c and 4d). Increasing the variations in SOM properties and microbial physiology from ±20% to ±50% substantially increased the variations of the simulated living microbial and enzyme pools. The simulated average annual mean living microbial and enzyme pools varied from 35.8 to 286.1 kg C ha⁻¹ and 0.19–4.97 kg C ha⁻¹across the scenarios in V50% (Figures 4c and 4d). Simulated living microbial pools were positively correlated with wheat residue return, amount of organic manure amendments, CAE, and initial SOC stock (P < 0.01; Table 3). However, relatively strong correlations (|r| > 0.3) appeared only for wheat residue return, amount of organic manure amendments, *CAE*, and rate of microbial maintenance respiration. Responses of simulated soil enzyme pools were similar to the responses of the simulated soil living microbial biomass (Equation 12). In addition, simulated soil enzyme pools were positively correlated with enzyme production was modeled as a constant faction of living microbial biomass (Equation 12). In addition, simulated soil enzyme pools were positively correlated with enzyme production rate and negatively correlated with enzyme decay rate (P < 0.01; Table 3).



3.3. Sensitivity of SOC Change, CO₂ flux, and Soil Microbial and Enzyme Pools

The most sensitive factor controlling the average annual SOC change was the amount of manure amendments, although the simulated SOC changes were also strongly regulated by the maximum decomposition rate of humus and microbial physiological parameters, primarily *CAE*, F_{MICtoDOC} , and enzyme production and decay rates under the typical changes in FMPs and ±20% variations in SOM properties and microbial physiology (Figure 6a). The simulated CO₂ fluxes from soil heterotrophic respiration were predominantly controlled by the rate of C input from manure amendments or wheat residue return at the study site (as demonstrated by high sensitivity index values for manure amendments and fraction of residue return, Figure 6c). The simulated soil living microbial and enzyme pools were largely regulated by both the C input rate from manure amendments or residue return and microbial physiology (Figures 6e and 6g). The microbial physiological parameters exerting relatively large impacts were CAE and $R_{\text{MicrobeMaintenance}}$ for the simulated living microbial pool and $R_{\text{EnzymeProduction}}$, $R_{\text{EnzymeDecay}}$, and CAE for the simulated enzyme pool.

Increasing the variations of the initial SOM stock, maximum SOM decomposition rate, and microbial physiology noticeably increased their influences on the simulated SOC changes as well as soil living microbial and enzyme pools. In comparison with the C input rate, CAE and $R_{\text{MicrobeMaintenance}}$ exerted larger impacts on the simulated soil living microbial pools and CAE, $R_{\text{MicrobeMaintenance}}$, $R_{\text{EnzymeProduction}}$, and $R_{\text{EnzymeDecay}}$ exerted larger impacts on the simulated soil enzyme pools (Figures 6f and 6h). However, their influences on the simulated SOC changes were similar to or smaller than the amount of manure amendments (Figure 6b). In addition, increasing the variations of these parameters only slightly increased their influence on simulated CO₂ fluxes from soil heterotrophic respiration (Figures 6c and 6d). Therefore, as expected, C input from organic manure or crop residue played an important role in regulating SOC change and CO₂ flux, even under the conditions with relatively large variations in SOM properties and microbial physiology.

4. Discussion

4.1. Model Performance

In this study, we developed a microbe-driven SOM decomposition model based on a traditional biogeochemical model, DNDC, and then investigated the impacts of FMPs, SOM, and microbial physiology on SOC changes and CO_2 fluxes from soil heterotrophic respiration. The developed model provides a framework to predict SOC changes and CO_2 fluxes by explicitly simulating both microbial regulations, impacts of plants, and common FMPs on soil C cycling.

The model was evaluated against the observed NEE and long-term SOC changes from the two winter wheat fields. In comparisons with the field observations, the model generally captured the magnitudes and seasonal dynamics of the daily NEE at the OK wheat field (Figure 3). In particular, the incorporation of the microbial processes on SOM decomposition and DOC uptake appreciably improved DNDC's simulation of the peak CO_2 fluxes induced by the wheat residue return (Figure 3) primarily because the modified model can capture priming effects of wheat residue inputs by explicitly simulating microbial activity (Blagodatsky et al., 2010), while the original DNDC could not simulate the priming effects due to missing relevant microbial dynamics. The *R* values between the improved DNDC's simulations and observations of daily NEE values were 0.78 and 0.82 in the first and second rotational years, respectively, and were higher than the corresponding values of 0.70 and 0.77 calculated based on simulations from the original DNDC. The improvements in the model's performance in simulating CO_2 fluxes from agricultural ecosystems. However, we note that adding the new processes also introduced new parameters that are not well-enough constrained by literature values and require calibration. Our parameter calibration was based only on long-term SOC simulations from the Rothamsted winter wheat NPK treatment, not the OK winter wheat field.

The modified model also captured the differences of the SOC changes under the three treatments with different C and N inputs at the Rothamsted site (Figure 2), suggesting that the microbe-driven SOM decomposition model can be used to quantify the long-term SOC changes for the studied winter wheat fields. However, the performance of simulating long-term SOC dynamics by the modified model is similar to that of the original DNDC in simulating SOC changes of the same fields (Li et al., 1994; Smith et al., 1997), even



though the original version is equipped with first-order kinetics to simulate SOM decomposition. Both the original and modified models captured the differences of SOC change among the different treatments. The RRMSE values (5%–8% among the treatments) calculated using the simulations from the modified DNDC were comparable with the RRMSE (7%–8%) based on the original DNDC simulations. Therefore, the incorporation of the microbial processes and activities on SOM decomposition in this study did not considerably improve the model's performance in capturing the observed SOC changes over decadal to century time scales. This may be because any missing microbial processes and activities of long-term SOM decomposition could be compensated by calibration of the SOM decomposition rates.

We note some discrepancies between the modeled results and field measurements. The modified DNDC model over-predicted CO, release rates following the wheat residue incorporation and slightly under-predicted CO, release rates in August and September in 2016 at the OK site. The over-predictions of the CO₂ release rates may have resulted from over-predictions of either litter availability or microbial growth and activity following the wheat residue incorporation, causing DNDC to predict relatively higher litter decomposition rates, DOC availability, and subsequently CO₂ production and release. Over-predicting initial litter decomposition could reduce the litter availability in the following August and September, and therefore be partially responsible for the under-predicted CO₂ release rates. The model also under-estimated CO, release rates in May and June in 2015. The under-estimations of the CO, fluxes could be due to under-estimation of wheat autotrophic respiration considering that soil heterotrophic respiration would be low because there was no wheat residue incorporation or tillage during this period. Further studies, and additional years of field observations, are needed to better explain the discrepancies between the simulations and observations. In addition, it should be noted that there are uncertainties in the eddy covariance measurements of NEE due to uncertainties associated with the instrument, source heterogeneity, gap-filling, and the turbulent nature of the transport process (Mauder et al., 2013; Richardson et al., 2006). These measurement uncertainties can also contribute to discrepancies between simulated and measured NEE. For example, random measurement errors were estimated from around 20%-80% of the NEE values at several croplands (Mauder et al., 2013; Richardson et al., 2006). These measurement uncertainties are larger than the discrepancies between the simulated and measured annual total NEE in this study.

4.2. Impacts on Soil Carbon Dynamics

The model predictions under different scenarios quantified the sensitivity of SOC change and CO_2 flux from soil heterotrophic respiration to changing FMPs, microbial physiology, and SOM property. In general, the simulations demonstrated that SOC changes were largely regulated by the C input from manure amendments and the soil CO_2 fluxes were predominantly controlled by the C inputs through manure amendments and wheat residue return (Figure 6). Simulated SOC changes were strongly correlated with the manure amendments, and simulated soil CO_2 fluxes were strongly correlated with C input rates through wheat residue return across the simulated scenarios (Table 3). These results are consistent with previous studies that have highlighted the importance of FMPs in controlling SOC change and CO_2 fluxes in agricultural ecosystems (e.g., McLauchlan, 2006).

Furthermore, in addition to the microbial physiology parameters, manure amendments and wheat residue return rates exerted large impacts on microbial activities (Figure 6; Blagodatskaya & Kuzyakov, 2013; Six et al., 2006). For example, the model predicted relatively low living microbial biomass under scenarios with low C inputs even though the microbial physiology parameters (e.g., high $V_{\text{MaxDOCuptake}}$ or CAE) were favorable for microbial growth. Although SOC changes were also regulated by parameters other than C inputs (e.g., CAE, F_{MICtoDOC} , and V_{Max}), particularly for the variations of $\pm 50\%$ of the baseline for these parameters, we note that the sensitivities depended on the variation ranges of the parameters (Figure 6). While the variations of FMPs were set to represent common changes in agricultural management, the $\pm 50\%$ variations for the microbial physiology might be overestimated for a specific cropland (i.e., the Rothamsted winter wheat field in this study). For example, the CAE was the most important microbial parameter regulating the SOC change and CO₂ flux (Figure 6) and the $\pm 50\%$ variation of the baseline value was generally larger than the CAE modifications investigated in the previous studies (e.g., Allison et al., 2010; Wang et al., 2013). Therefore, the importance of CAE in regulating the SOC change and CO₂ flux under the $\pm 50\%$ variations might be overestimated. Based on the large impacts of FMPs in governing SOC change, CO₂



flux, and microbial activities as predicted in this study and reported in other studies (e.g., Blagodatskaya & Kuzyakov, 2013; Böhme et al., 2005), we conclude that SOC dynamics might be largely regulated by FMPs in agricultural ecosystems, although microbes played a central role in decomposing SOM.

The model also predicted differential impacts on SOC changes and CO_2 fluxes of different FMPs through incorporating processes to characterize these FMPs. For example, different responses of SOC changes and CO_2 fluxes were predicted between the applications of FYM and wheat residue. Consistent with the field measurements at the Rothamsted site, the model simulated that most C inputs from wheat straw were released as CO_2 fluxes and a larger fraction of added manure C was retained in the soil compared to C inputs from wheat straw (Figure 5; Johnston et al., 2009). As a result, the impact on simulated SOC changes of manure amendment was larger than the impact of wheat straw return (Figure 6). The predicted higher efficiencies of organic carbon retention by applying manure were primarily attributed to the different qualities of organic matter. In comparison with the wheat straw, which had a C/N ratio of about 65, the organic manure had a lower C/N ratio of 12.5 (Table 2). Therefore, for a given amount of organic matter input, the model predicted more manure C converted into microbial biomass during decomposition due to lower N limitations for microbial growth, and thereby predicted more C retained in the soil derived from microbial death. In addition, the model partitions a fraction of manure into the humads pool (Li et al., 2012) that could be directly decomposed into the passive humus pool with low decomposition rates.

We note that microbial physiology may exert considerable impacts on SOC changes and soil CO, fluxes. Simulated SOC increase rates tended to increase with increasing CAE and decreasing F_{MICtoDOC} if the FMPs, SOM, and other microbial physiology parameters were not changed (Figure 5). Simulated SOC changes were positively correlated with CAE (P < 0.01) and negatively correlated with F_{MICIDDOC} (P < 0.01) even under conditions with relatively large variations in C input (0.0–1.0 for fraction of returned crop residue, and 0-2,800 kg C ha⁻¹ year⁻¹ for organic manure input) (Table 3). In addition, the sensitivity of simulated SOC changes to changes in CAE or F_{MICtoDOC} were comparable with that to changes in wheat residue return and manure application rate under the V50 scenarios (Figure 6b). Simulated annual CO, fluxes tended to be lower in scenarios with higher F_{MICtoDOC} and lower CAE if the other factors were not changed (Figure 5). Annual CO₂ fluxes also significantly correlated with these two parameters under the V50 scenarios (Table 3), indicating appreciable impacts of microbial physiology on soil heterotrophic respiration in agricultural ecosystems. These results are consistent with previous studies that have suggested the importance of microbial physiology in controlling SOC change and CO, fluxes (e.g., Liang et al., 2019; Six et al., 2006; Wang et al., 2021). Furthermore, the enzyme production and decay rates exerted comparable impacts with manure amendments in regulating the SOC changes under $\pm 50\%$ variations in microbial physiology parameters (Figure 6). Therefore, the function of enzymes in catalyzing SOC decomposition is considerable, although the enzyme concentrations are very small compared to other C pools. These results highlight the central role of soil microbes in converting the C inputs to resistant SOC, and suggest that microbial physiology and activities need to be considered when assessing SOC changes even in agricultural ecosystems where FMPs usually play a major role in controlling SOC dynamics.

4.3. Integrating Soil Microbial Dynamics Into Agricultural Ecosystem Models

Modeling microbial-mediated SOM dynamics in agricultural ecosystems is in an early stage, and much additional work is required for a more complete framework of all of the processes involved. As shown in this and other studies (e.g., Bastian et al., 2009; Böhme et al., 2005; Bossio et al., 1998), microbial biomass and activity are strongly regulated by crop residue incorporation and other FMPs, which stresses the importance of correct simulations of crop dynamics and parameterization of FMPs when predicting responses of microbial activities to environmental changes and their influences on SOM decomposition and CO_2 fluxes in agricultural ecosystems. However, most microbial models that investigate impacts of soil microbes on SOM turnover are based on static or prescribed C input (e.g., Abramoff et al., 2017; Schimel & Weintraub, 2003; Sulman et al., 2018). Therefore, biases and uncertainty in predicting SOM dynamics may result from neglecting crop dynamics and associated changes in crop residue return. Furthermore, the impacts of FMPs on microbial and enzyme activities are large but highly uncertain (Burns et al., 2013). Therefore, further studies need to be performed to quantify how FMPs affect microbial dynamics, how this in turn affects veg-



etation and SOM dynamics, and how the interactions among FMPs, soil microbes, crop growth, and SOM turnover are best incorporated into process-based biogeochemical models.

Another uncertainty in the DNDC model framework is the representation of SOM pools. The multiple SOM pools in DNDC were designed to empirically represent decay rates of a wide variety of organic matter compounds, with the multiple SOM pool dynamics simulating overall actual SOM dynamics (Li et al., 1992a). The approach of multiple SOM pools is applicable for predicting SOC dynamics under varied FMPs (e.g., Figure 2; Li et al., 1994, 1997), and may be necessary for simulating SOM dynamics in agricultural ecosystems where organic amendments and SOM consists of materials at various stages of decomposition and that have different characteristics (De Graaff et al., 2010; Lal, 2018). However, the pools based on decay rates are conceptual and difficult to relate to measurements (Schmidt et al., 2011), although this approach may be more informative than the model with a single SOC pool (e.g., Allison et al., 2010). In addition, the modified DNDC model presented here simplifies the processes of microbial and enzyme dynamics. For example, the model does not differentiate microbial communities responsible for SOM decomposition (i.e., it simulates a single microbial pool) in order to minimize the number of poorly constrained microbial parameters and to maintain the applicability of DNDC to agricultural ecosystems with a wide range of management practices. Further steps to incorporate microbial communities and responses of microbial and enzyme activities to substrate chemistry and microbial stoichiometry (Sinsabaugh et al., 2008) into mathematical models will be required for a complete model framework.

High uncertainty in parameters related to soil microbes and SOM decomposition were also detected. Several new parameters were estimated by calibrating the simulations against the field observations of SOC changes under the NPK treatment due to a lack of specific parameter values (Table 1). Most of the calibrated parameters were comparable with published values. For example, the maximum rates of decomposition of various SOM pools and DOC uptake were within the reported variations (e.g., Wang et al., 2013). However, little information can be found for some parameters, such as the half-saturation constants in the Michaelis-Menten equations for calculating the decomposition of the SOM pools, because these parameters could be variable across different environmental conditions and are poorly constrained by observations (Moorhead & Sinsabaugh, 2006). Evaluating the model against more observations of microbial activities and C dynamics from different locations and under different environmental conditions should help to constrain uncertain parameters. Furthermore, it is hard to estimate the plausible variations of the parameters related to soil microbes and SOM decomposition for a specific ecosystem or site (e.g., the winter wheat system investigated in this study). Insufficient constraints on the parameters suggest that further studies are required to identify plausible ranges for parameters regarding microbial dynamics and SOM decomposition, and how these parameters depend on environmental conditions. Our efforts incorporating microbial regulations into a traditional biogeochemical model and investigating responses of SOC change and CO, flux to varied FMPs, SOM, and microbial physiology provide critical guidance on identifying processes/parameters for which soil C stocks and CO₂ fluxes are strongly sensitive.

5. Conclusions

We developed a microbial-mediated SOM decomposition model, based on a biogeochemical model (DNDC), to simulate C dynamics in agricultural ecosystems. The model simulates decomposition of SOM and external organic matter, soil respiration, and SOC formation by explicitly simulating microbial and enzyme dynamics and their controls on SOM decomposition, in addition to simulating impacts on C dynamics of climate, soil properties, crop, and common FMPs. The model was evaluated against field observations of NEE and SOC change in two winter wheat systems to assess its performance in predicting both CO_2 fluxes and SOC changes. The model successfully captured seasonal variations of NEE and SOC changes under different FMPs. The incorporation of microbial processes improved the model's performance in simulating peak CO_2 fluxes induced by the wheat residue return primarily because the improved model captured priming effects of wheat residue inputs. We applied the model to investigate impacts of microbial physiology, SOM, and FMPs on soil microbial activity, SOC change, and CO_2 fluxes. The results show that C input through crop residue or manure drove microbial activity and predominantly regulated the soil CO_2 fluxes, and that manure amendment largely regulated SOC change. Microbial physiology also exerted considerable impacts



on the microbial activities, SOC changes, and soil CO_2 fluxes, emphasizing the necessity of considering microbial physiology and activities when assessing soil C dynamics in agricultural ecosystems where FMPs usually play a major role in controlling soil C.

Appendix A: Key Equations to Simulate SOM Decomposition, and Dynamics of Soil Organic Matter (SOM), Dissolved Organic Carbon (DOC), Dissolved Organic Nitrogen (DON), Soil Microbe Biomass, and Enzyme Amount

Appendix A1: Decomposition of SOM Pools

Decomposition of very liable litter

$$D_{\text{Litter}_vl} = V \max_{\text{Litter}_vl} \times \text{Enzyme} \times \frac{\text{Litter}_vl}{K_{\text{Litter}_vl} + \text{Litter}_vl} \times f(T) \times f(W) \times f(\text{ST})$$
(A1)

Decomposition of liable litter

$$D_{\text{Litter}_{l}} = V \max_{\text{Litter}_{l}} \times \text{Enzyme} \times \frac{\text{Litter}_{l}}{K_{\text{Litter}_{l}} + \text{Litter}_{l}} \times f(T) \times f(W) \times f(\text{ST})$$
(A2)

Decomposition of resistant litter

$$D_{\text{Litter}_r} = V \max_{\text{Litter}_r} \times \text{Enzyme} \times \frac{\text{Litter}_r}{K_{\text{Litter}_r} + \text{Litter}_r} \times f(T) \times f(W) \times f(\text{ST})$$
(A3)

Total decomposition of litter

$$D_{\text{Litter}} = D_{\text{Litter}_vl} + D_{\text{Litter}_l} + D_{\text{Litter}_r}$$
(A4)

Decomposition of liable humads

$$D_{\text{Humads}_l} = V \max_{\text{Humads}_l} \times \text{Enzyme} \times \frac{\text{Humads}_l}{K_{\text{Humads}_l} + \text{Humads}_l} \times f(T) \times f(W) \times f(\text{ST})$$
(A5)

Decomposition of resistant humads

$$D_{\text{Humads}_r} = V \max_{\text{Humads}_r} \times \text{Enzyme} \times \frac{\text{Humads}_r}{K_{\text{Humads}_r} + \text{Humads}_r} \times f(T) \times f(W) \times f(\text{ST})$$
(A6)

Decomposition of humus

$$D_{\text{Humus}} = V \max_{\text{Humus}} \times \text{Enzyme} \times \frac{\text{Humus}}{K_{\text{Humus}} + \text{Humus}} \times f(T) \times f(W) \times f(\text{ST})$$
(A7)

The Arrhenius equation

$$f(T) = \exp\left[-\frac{Ea}{R} \times \left(\frac{1}{T} - \frac{1}{T_{ref}}\right)\right]$$
(A8)

Impact of soil water on SOM decomposition

$$f(SW) = -2.8516 \times SW^3 + 1.4936 \times SW^2 + 1.7699 \times SW - 0.0301$$
(A9)

Impact of clay content on SOM decomposition

$$f(ST) = 0.5 \times Clay^{-0.471}$$
 (A10)



Appendix A2: Dynamics of SOM, DOC, and DON

Dynamics of litter pools

$$\frac{d\text{Litter}}{dt} = \text{Input}_{\text{Litter}} - D_{\text{Litter}}$$
(A11)

Dynamics of humads pools

$$\frac{d\text{Humads}}{dt} = \text{Death}_{\text{Microbes}} \times (1.0 - F_{\text{MICToDOC}}) - D_{\text{Humads}}$$
(A12)

Dynamics of humus pool

$$\frac{d\text{Humus}}{dt} = D_{\text{Humads}} \times \left(1.0 - F_{\text{HumadsToDOC}}\right) - D_{\text{Humus}}$$
(A13)

Dynamics of DOC pool

$$\frac{d\text{DOC}}{dt} = \text{DOC}_{\text{Manure}} + \text{DOC}_{\text{Root}} + D_{\text{Litter}} + D_{\text{Humads}} \times F_{\text{HumadsToDOC}} + D_{\text{Humus}} + \text{Death}_{\text{Microbe}} \times F_{\text{MICToDOC}} + D_{\text{Enzyme}} - \text{DOC}_{\text{Uptake}} - \text{DOC}_{N} - \text{DOC}_{M} - \text{DOC}_{\text{Leaching}}$$
(A14)

Potential DOC uptake

$$DOC_{PUptake} = V \max_{DOCUptake} \times Microbe \times \frac{DOC}{K_{DOC} + DOC} \times \frac{O_2}{K_{O_2} + O_2} \times f(T)$$
(A15)

Actual DOC uptake

$$DOC_{Uptake} = DOC_{PUptake}$$
 If No DOC and DON *l*imitation
DOC_{Uptake} = DOC If limited by soil DOC or DON availability (A16)

Dynamics of DON pool

$$\frac{d\text{DON}}{dt} = \text{DON}_{\text{Manure}} + \text{DON}_{\text{Root}} + \frac{D_{\text{Litter}}}{\text{CN}_{\text{Litter}}} + \frac{D_{\text{Humads}} \times F_{\text{HumadsToDOC}}}{\text{CN}_{\text{Humads}}} + \frac{D_{\text{Humus}}}{\text{CN}_{\text{Humus}}} + \frac{D_{\text{enzyme}}}{\text{CN}_{\text{Microbe}}} - \text{DON}_{\text{Uptake}} - \text{DON}_{\text{Leaching}}$$
(A17)

Appendix A3: Dynamics of Soil Microbes and Enzyme

Dynamics of soil living microbes

$$\frac{d\text{Microbe}}{dt} = \text{Growth}_{\text{Microbe}} - \text{Death}_{\text{Microbe}} - P_{\text{Enzyme}}$$
(A18)

Growth of microbial biomass

$$Growth_{Microbe} = \left(DOC_{Uptake} - M_{Maintenance}\right) \times CAE \quad If \frac{DN}{N_{Demand}} \ge 1.0$$

$$Growth_{Microbe} = \left(DOC_{Uptake} - M_{Maintenance}\right) \times CAE \times \frac{DN}{N_{Demand}} If \frac{DN}{N_{Demand}} < 1.0$$
(A19)

Nitrogen demand for microbial growth

$$N_{\text{Demand}} = \left(\text{DOC}_{\text{Uptake}} - M_{\text{Maintenance}}\right) \times \text{CAE} / CN_{\text{Microbe}}$$
(A20)



Microbial maintenance respiration		
	$M_{\text{Maintenance}} = R_{\text{Microbe Maintenance}} \times \text{Microbe}$	(A21)
Dissolved nitrogen		
	DN = DON + MN	(A22)
Microbial mortality		
	$\text{Death}_{\text{Microbe}} = R_{\text{Microbe Death}} \times \text{Microbe}^2$	(A23)
Enzyme dynamic		
	$\frac{d\text{Enzyme}}{dt} = P_{\text{Enzyme}} - D_{\text{Enzyme}}$	(A24)
Enzyme production		
	$P_{\rm Enzyme} = R_{\rm EnzymeProduction} \times {\rm Microbe}$	(A25)
Enzyme decay		
	$D_{\text{Enzyme}} = R_{\text{Enzyme Decay}} \times \text{Enzyme}$	(A26)

Appendix B: Definitions of Variables Listed in Appendix A

Variable	Definition, unit
CAE	Maximum microbial carbon assimilation efficiency
Clay	Soil clay content, %
CN_{Litter} , CN_{Humads} , CN_{Humus} , $CN_{Microbe}$, CN_{Enzyme}	C:N ratios of litters, humads, humus, soil microbes, and enzyme respectively
$D_{\text{Litter'}} D_{\text{Litter_vP}} D_{\text{Litter_r'}} D_{\text{Humads_P}} D_{\text{Humads_r'}} D_{\text{Humus}}$	Decomposition rates of litters, very liable litter, liable litter, resistant litter, liable humads, resistant humads, and humus, respectively
D _{Enzyme}	Enzyme decay rate, h ⁻¹
Death _{Microbe}	Mortality of microbial biomass
DN	Dissolved nitrogen, including both DON and mineral N
DOC	Concentration of DOC, mg C g ⁻¹ soil
DOC _{Leaching}	DOC transferred out of soils through water leaching
DOC _M	DOC consumed through methanogenesis
DOC _{Manure}	DOC input from organic manure
DOC _N	DOC consumed through denitrification
DOC _{Root}	DOC input from root exudation
DOC _{Uptake}	DOC uptake by microbes
DON	Concentration of DON, mg N g ⁻¹ soil
DON _{Leaching}	DON transferred out of soils through water leaching
DON _{Manure}	DON input through organic manure
DON _{Root}	DON input through root exudation
DON _{Uptake}	DON uptake by microbes
$Ea_{\text{Litter_}\nuP} Ea_{\text{Litter_}P} Ea_{\text{Litter_}P} Ea_{\text{Humads_}P} Ea_{\text{Humads_}P} Ea_{\text{Humads_}P}$	Activation energies in the Arrhenius equation for decomposition of very labile litter, labile litter, resistant litter, labile humads, resistant humads, and humus, respectively, KJ mol ⁻¹
$F_{ m HumadsToDOC}$	Fractions of decomposed humads that allocated to DOC



Variable	Definition, unit
F _{MICtoDOC}	Fraction of dead microbial biomass that allocated to DOC
Growth _{Microbe}	Growth of microbial biomass
Humads_l	Concentration of labile humads, mg C g ⁻¹ soil
Humads_r	Concentration of resistant humads, mg C g ⁻¹ soil
Humus	Concentration of humus, mg C g ⁻¹ soil
Input _{Litter}	Litter input from crop residue
$K_{\text{Litter_}\nu p} \; K_{\text{Litter_}p} \; K_{\text{Litter_}p} \; K_{\text{Humads_}p} \; K_{\text{Humads_}p} \; K_{\text{Humads_}p}$	Michaelis half-saturation constant for decomposition of very labile litter, labile litter, resistant litter, labile humads, resistant humads, and humus, respectively, mg C g ⁻¹ soil
K _{DOC}	Michaelis DOC half-saturation constant for DOC uptake, mg C $\rm g^{-1}$ soil
<i>K</i> ₀₂	Michaelis $\rm O_2$ half-saturation constant for DOC uptake, mmol cm $^{-3}$
Litter_vl	Concentration of very labile litter, mg C g^{-1} soil
Litter_l	Concentration of labile litter, mg C g ⁻¹ soil
Litter_r	Concentration of resistant litter, mg C g ⁻¹ soil
Microbe	Concentration of soil living microbes, mg C g ⁻¹ soil
MN	Soil mineral nitrogen
$N_{ m Demand}$	Potential N required for microbe growth
0 ₂	Concentration of soil oxygen, mmol cm ⁻³ .
P _{Enzyme}	Enzyme production
R	The gas constant, 8.314 J K ⁻¹ mol ⁻¹
R _{MicrobeMaintenance}	Rate of microbial maintenance respiration, h ⁻¹
R _{MicrobeDeath}	Microbial turnover rate, h ⁻¹
R _{EnzymeProduction}	Enzyme production rate, h ⁻¹
R _{EnzymeDecay}	Enzyme decay rate, h ⁻¹
SW	Soil moisture in water- filled fraction of total porosity
Т	Soil temperature, °C
$V \text{max}_{\text{Litter}_\nu^p} V \text{max}_{\text{Litter}_\prime} V \text{max}_{\text{Humads}_\prime} V V \text{max}_{\text{Humads}_\prime} V V V V V V V V V V V V V V V V V V V$	Maximum decomposition rate of very labile litter, labile litter, resistant litter, labile humads, resistant humads, and humus, respectively, at reference temperature, mg C (mg Enzyme C) ⁻¹ hr ⁻¹
Vmax _{DOCuptake}	Maximum uptake rate of DOC at a reference temperature of 20°C, mg C (mg microbe C) ⁻¹ hr ⁻¹

Data Availability Statement

The DNDC model, model input files, and all data used in this study are archived at the figshare repository (https://doi.org/10.6084/m9.figshare.15131976.v2).

Acknowledgments

This research was supported by USDA National Institute of Food and Agriculture (Grant No. 2016-68002-24967) and National Aeronautics and Space Administration (NASA) Interdisciplinary Research in Earth Science (Grant No. NNX17AK10G). The authors thank the Rothamsted Research for providing the long-term soil organic carbon data.

References

Abramoff, R. Z., Davidson, E. A., & Finzi, A. C. (2017). A parsimonious modular approach to building a mechanistic belowground carbon and nitrogen model. *Journal of Geophysical Research: Biogeosciences*, *122*(9), 2418–2434. https://doi.org/10.1002/2017JG003796

Allison, S. D., Wallenstein, M. D., & Bradford, M. A. (2010). Soil-carbon response to warming dependent on microbial physiology. *Nature Geoscience*, *3*(5), 336–340. https://doi.org/10.1038/ngeo846

Amelung, W., Bossio, D., de Vries, W., Kögel-Knabner, I., Lehmann, J., Amundson, R., et al. (2020). Towards a global-scale soil climate mitigation strategy. *Nature Communications*, 11, 1–10. https://doi.org/10.1038/s41467-020-18887-7

Bajgain, R., Xiao, X., Basara, J., Wagle, P., Zhou, Y., Mahan, H., et al. (2018). Carbon dioxide and water vapor fluxes in winter wheat and tallgrass prairie in central Oklahoma. *Science of the Total Environment*, 644, 1511–1524. https://doi.org/10.1016/j.scitotenv.2018.07.010
 Bastian, F., Bouziri, L., Nicolardot, B., & Ranjard, L. (2009). Impact of wheat straw decomposition on successional patterns of soil microbial community structure. *Soil Biology and Biochemistry*, 41(2), 262–275. https://doi.org/10.1016/j.soilbio.2008.10.024

Blagodatskaya, E., & Kuzyakov, Y. (2013). Active microorganisms in soil: Critical review of estimation criteria and approaches. Soil Biology and Biochemistry, 67, 192–211. https://doi.org/10.1016/j.soilbio.2013.08.024



- Blagodatsky, S., Blagodatskaya, E., Yuyukina, T., & Kuzyakov, Y. (2010). Model of apparent and real priming effects: Linking microbial activity with soil organic matter decomposition. Soil Biology and Biochemistry, 42(8), 1275–1283. https://doi.org/10.1016/j.soilbio.2010.04.005 Böhme, L., Langer, U., & Böhme, F. (2005). Microbial biomass, enzyme activities and microbial community structure in two European
- long-term field experiments. Agriculture, Ecosystems & Environment, 109(1–2), 141–152. https://doi.org/10.1016/j.agee.2005.01.017
 Bossio, D. A., Scow, K. M., Gunapala, N., & Graham, K. J. (1998). Determinants of soil microbial communities: Effects of agricultural management, season, and soil type on phospholipid fatty acid profiles. Microbial Ecology, 36(1), 1–12. https://doi.org/10.1007/s002489900087
- Burns, R. G., DeForest, J. L., Marssen, J., Sinsabaugh, R. L., Stromberger, M. E., Wallenstein, M. D., et al. (2013). Soil enzymes in a changing environment: Current knowledge and future directions. Soil Biology and Biochemistry, 58, 216–234. https://doi.org/10.1016/j. soilbio.2012.11.009
- De Graaff, M. A., Classen, A. T., Castro, H. F., & Schadt, C. W. (2010). Labile soil carbon inputs mediate the soil microbial community composition and plant residue decomposition rates. New Phytologist, 188(4), 1055–1064. https://doi.org/10.1111/j.1469-8137.2010.03427.x
- Deng, J., Li, C., Burger, M., Horwath, W. R., Smart, D., Six, J., et al. (2018). Assessing short-term impacts of management practices on N₂O emissions from diverse Mediterranean agricultural ecosystems using a biogeochemical model. *Journal of Geophysical Research: Biogeosciences*, 123(5), 1557–1571. https://doi.org/10.1029/2017JG004260
- Deng, J., Li, C., Frolking, S., Zhang, Y., Backstrand, K., & Crill, P. (2014). Assessing effects of permafrost thaw on C fluxes based on multiyear modeling across a permafrost thaw gradient at Stordalen, Sweden. *Biogeosciences*, 11, 4753–4770. https://doi.org/10.5194/ bg-11-4753-2014
- Deng, J., McCalley, C. K., Frolking, S., Chanton, J., Crill, P., Varner, R., et al. (2017). Adding stable carbon isotopes improves model representation of the role of microbial communities in peatland methane cycling. *Journal of Advances in Modeling Earth Systems*, 9(2), 1412–1430. https://doi.org/10.1002/2016MS000817
- Deng, J., Xiao, J., Ouimette, A., Zhang, Y., Sanders-DeMott, R., Frolking, S., & Li, C. (2020). Improving a biogeochemical model to simulate surface energy, greenhouse gas fluxes, and radiative forcing for different land use types in northeastern United States. *Global Biogeochemical Cycles*, 34(8), e2019GB006520. https://doi.org/10.1029/2019GB006520
- Deng, J., Zhu, B., Zhou, Z., Zheng, X., Li, C., Wang, T., & Tang, J. (2011). Modeling nitrogen loadings from agricultural soils in Southwest China with modified DNDC. Journal of Geophysical Research: Biogeosciences, 116, G02020. https://doi.org/10.1029/2010JG001609
- Dignac, M. F., Derrien, D., Barre, P., Barot, S., Cécillon, L., Chenu, C., et al. (2017). Increasing soil carbon storage: Mechanisms, effects of agricultural practices and proxies. A review. Agronomy for Sustainable Development, 37(2), 14. https://doi.org/10.1007/s13593-017-0421-2
- Fischer, M. L., Torn, M. S., Billesbach, D. P., Doyle, G., Northup, B., & Biraud, S. C. (2012). Carbon, water, and heat flux responses to experimental burning and drought in a tallgrass prairie. Agricultural and Forest Meteorology, 166, 169–174. https://doi.org/10.1016/j. agrformet.2012.07.011
- Gao, Q., Wang, G., Xue, K., Yang, Y., Xie, J., Yu, H., et al. (2020). Stimulation of soil respiration by elevated CO₂ is enhanced under nitrogen limitation in a decade-long grassland study. *Proceedings of the National Academy of Sciences*, 117, 33317–33324. https://doi.org/10.1073/ pnas.2002780117
- Georgiou, K., Abramoff, R. Z., Harte, J., Riley, W. J., & Torn, M. S. (2017). Microbial community-level regulation explains soil carbon responses to long-term litter manipulations. *Nature Communications*, 8(1), 1223. https://doi.org/10.1038/s41467-017-01116-z
- Gilhespy, S. L., Anthony, S., Cardenas, L., Chadwick, D., del Prado, A., Li, C., et al. (2014). First 20 years of DNDC (DeNitrification DeComposition): Model evolution. *Ecological Modelling*, 292(24), 51–62. https://doi.org/10.1016/j.ecolmodel.2014.09.004
- Giltrap, D. L., Li, C., & Saggar, S. (2010). DNDC: A process-based model of greenhouse gas fluxes from agricultural soils. Agriculture, Ecosystems & Environment, 136(3-4), 292–300. https://doi.org/10.1016/j.agee.2009.06.014
- Guo, X., Gao, Q., Yuan, M., Wang, G., Zhou, X., Feng, J., et al. (2020). Gene-informed decomposition model predicts lower soil carbon loss due to persistent microbial adaptation to warming. *Nature Communications*, 11, 4897. https://doi.org/10.1038/s41467-020-18706-z
- He, Y., Yang, J., Zhuang, Q., Harden, J. W., McGuire, A. D., Liu, Y., et al. (2015). Incorporating microbial dormancy dynamics into soil decomposition models to improve quantification of soil carbon dynamics of northern temperate forests. *Journal of Geophysical Research: Biogeosciences*, 120(12), 2596–2611. https://doi.org/10.1002/2015JG003130
- Helton, J. C., & Davis, F. J. (2003). Latin hypercube sampling and the propagation of uncertainty in analyses of complex systems. *Reliability Engineering & System Safety*, 81(1), 23–69. https://doi.org/10.1016/S0951-8320(03)00058-9
- Huang, Y., Guenet, B., Ciais, P., Janssens, I. A., Soong, J. L., Wang, Y., et al. (2018). ORCHIMIC (v1.0), a microbe-mediated model for soil organic matter decomposition. *Geoscientific Model Development*, 11(6), 2111–2138. https://doi.org/10.5194/gmd-11-2111-2018
- IPCC. (2013). Climate change 2013: The physical science basis. Contribution of working group I to the fifth assessment report of the Intergovernmental Panel on Climate Change. Cambridge University Press.
- Johnston, A. E., Poulton, P. R., & Coleman, K. (2009). Soil organic matter: Its importance in sustainable agriculture and carbon dioxide fluxes. Advances in Agronomy, 101, 1–57. https://doi.org/10.1016/S0065-2113(08)00801-8
- Kallenbach, C. M., Frey, S. D., & Grandy, A. S. (2016). Direct evidence for microbial-derived soil organic matter formation and its ecophysiological controls. *Nature Communications*, 7, 13630. https://doi.org/10.1038/ncomms13630
- Lal, R. (2004). Soil carbon sequestration impacts on global climate change and food security. *Science*, 304(5677), 1623–1627. https://doi.org/10.1126/science.1097396
- Lal, R. (2008). Soils and sustainable agriculture. A review. Agronomy for Sustainable Development, 28(1), 57–64. https://doi.org/10.1051/ agro:2007025
- Lal, R. (2018). Digging deeper: A holistic perspective of factors affecting soil organic carbon sequestration in agroecosystems. *Global Change Biology*, 24(8), 3285–3301. https://doi.org/10.1111/gcb.14054
- Lehmann, J., & Kleber, M. (2015). The contentious nature of soil organic matter. *Nature*, 528(7580), 60–68. https://doi.org/10.1038/ nature16069
- Li, C. (2000). Modeling trace gas emissions from agricultural ecosystems. Nutrient Cycling in Agroecosystems, 58(1–3), 259–276. https://doi. org/10.1023/A:1009859006242
- Li, C. (Ed.). (2016). Biogeochemistry: Scientific fundamentals and modeling approach. Tsinghua University Press.
- Li, C., Farahbakhshazad, N., Jaynes, D. B., Dinnes, D. L., Salas, W., & McLaughin, D. (2006). Modeling nitrate leaching with a biogeochemical model modified based on observations in a row-crop field in Iowa. *Ecological Modelling*, 196, 116–130. https://doi.org/10.1016/j. ecolmodel.2006.02.007
- Li, C., Frolking, S., Crocker, G. J., Grace, P. R., Klír, J., Körchens, M., & Poulton, P. R. (1997). Simulating trends in soil organic carbon in long-term experiments using the DNDC model. *Geoderma*, 81(1–2), 45–60. https://doi.org/10.1016/S0016-7061(97)00080-3
- Li, C., Frolking, S., & Frolking, T. A. (1992a). A model of nitrous oxide evolution from soil driven by rainfall events: 1. Model structure and sensitivity. Journal of Geophysical Research: Atmospheres, 97(D9), 9759–9776. https://doi.org/10.1029/92JD00509



- Li, C., Frolking, S., & Frolking, T. A. (1992b). A model of nitrous oxide evolution from soil driven by rainfall events: 2. Model applications. Journal of Geophysical Research: Atmospheres, 97(D9), 9777–9783. https://doi.org/10.1029/92JD00510
- Li, C., Frolking, S., & Harriss, R. (1994). Modeling carbon biogeochemistry in agricultural soils. Global Biogeochemical Cycles, 8(3), 237– 254. https://doi.org/10.1029/94GB00767
- Li, C., Salas, W., Zhang, R., Krauter, C., Rotz, A., & Mitloehner, F. (2012). Manure-DNDC: A biogeochemical process model for quantifying greenhouse gas and ammonia emissions from livestock manure systems. *Nutrient Cycling in Agroecosystems*, 93(2), 163–200. https:// doi.org/10.1007/s10705-012-9507-z
- Liang, C., Amelung, W., Lehmann, J., & Kästner, M. (2019). Quantitative assessment of microbial necromass contribution to soil organic matter. *Global Change Biology*, 25(11), 3578–3590. https://doi.org/10.1111/gcb.14781
- Liu, C., Lu, M., Cui, J., Li, B., & Fang, C. (2014). Effects of straw carbon input on carbon dynamics in agricultural soils: A meta-analysis. *Global Change Biology*, 20(5), 1366–1381. https://doi.org/10.1111/gcb.12517
- Maillard, É., & Angers, D. A. (2014). Animal manure application and soil organic carbon stocks: A meta-analysis. Global Change Biology, 20(2), 666–679. https://doi.org/10.1111/gcb.12438
- Manzoni, S., Taylor, P., Richter, A., Porporato, A., & Ågren, G. I. (2012). Environmental and stoichiometric controls on microbial carbon-use efficiency in soils. *New Phytologist*, 196(1), 79–91. https://doi.org/10.1111/j.1469-8137.2012.04225.x
- Mauder, M., Cuntz, M., Drüe, C., Graf, A., Rebmann, C., Schmid, H. P., et al. (2013). A strategy for quality and uncertainty assessment of long-term eddy-covariance measurements. *Agricultural and Forest Meteorology*, 169(15), 122–135. https://doi.org/10.1016/j. agrformet.2012.09.006
- McLauchlan, K. (2006). The nature and longevity of agricultural impacts on soil carbon and nutrients: A review. *Ecosystems*, 9(8), 1364–1382. https://doi.org/10.1007/s10021-005-0135-1
- Miltner, A., Bombach, P., Schmidt-Brücken, B., & Kästner, M. (2012). SOM genesis: Microbial biomass as a significant source. *Biogeochemistry*, 111(1–3), 41–55. https://doi.org/10.1007/s10533-011-9658-z
- Molina, J. A. E., Clapp, C. E., Shaffer, M. J., Chichester, F. W., & Larson, W. E. (1983). NCSOIL, a model of nitrogen and carbon transformations in soil: Description, calibration, and behavior. Soil Science Society of America Journal, 47(1), 85–91. https://doi.org/10.2136/ sssaj1983.03615995004700010017x
- Moorhead, D. L., & Sinsabaugh, R. L. (2006). A theoretical model of litter decay and microbial interaction. *Ecological Monographs*, 76(2), 151–174. https://doi.org/10.1890/0012-9615(2006)076[0151:ATMOLD]2.0.CO;2
- Moriasi, D. N., Arnold, J. G., Van Liew, M. W., Bingner, R. L., Harmel, R. D., & Veith, T. L. (2007). Model evaluation guidelines for systematic quantification of accuracy in watershed simulations. *Transactions of the ASABE*, 50(3), 885–900. https://doi.org/10.13031/2013.23153
- Parton, W. J., Gutmann, M. P., Merchant, E. R., Hartman, M. D., Adler, P. R., McNeal, F. M., & Lutz, S. M. (2015). Measuring and mitigating agricultural greenhouse gas production in the US Great Plains, 1870–2000. Proceedings of the National Academy of Sciences, 112(34), E4681–E4688. https://doi.org/10.1073/pnas.1416499112
- Paustian, K., Lehmann, J., Ogle, S., Reay, D., Robertson, G. P., & Smith, P. (2016). Climate-smart soils. Nature, 532, 49–57. https://doi. org/10.1038/nature17174
- Powlson, D. S., Bhogal, A., Chambers, B. J., Coleman, K., Macdonald, A. J., Goulding, K. W. T., & Whitmore, A. P. (2012). The potential to increase soil carbon stocks through reduced tillage or organic material additions in England and Wales: A case study. Agriculture, Ecosystems & Environment, 146(1), 23–33. https://doi.org/10.1016/j.agee.2011.10.004
- Prommer, J., Walker, T. W., Wanek, W., Braun, J., Zezula, D., Hu, Y., et al. (2020). Increased microbial growth, biomass, and turnover drive soil organic carbon accumulation at higher plant diversity. *Global Change Biology*, 26(2), 669–681. https://doi.org/10.1111/gcb.14777
- Richardson, A. D., Hollinger, D. Y., Burba, G. G., Davis, K. J., Flanagan, L. B., Katul, G. G., et al. (2006). A multi-site analysis of random error in tower-based measurements of carbon and energy fluxes. *Agricultural and Forest Meteorology*, 136(1–2), 1–18. https://doi. org/10.1016/j.agrformet.2006.01.007
- Saltelli, A., Ratto, M., Andres, T., Campolongo, F., Cariboni, J., Gatelli, D., Saisana, M., & Tarantola, S. (Eds.). (2008). Global sensitivity analysis: The primer. John Wiley.
- Schimel, J. P., & Weintraub, M. N. (2003). The implications of exoenzyme activity on microbial carbon and nitrogen limitation in soil: A theoretical model. Soil Biology and Biochemistry, 35(4), 549–563. https://doi.org/10.1016/S0038-0717(03)00015-4
- Schmidt, M. W., Torn, M. S., Abiven, S., Dittmar, T., Guggenberger, G., Janssens, I. A., et al. (2011). Persistence of soil organic matter as an ecosystem property. *Nature*, 478(7367), 49–56. https://doi.org/10.1038/nature10386
- Sinsabaugh, R. L., Lauber, C. L., Weintraub, M. N., Ahmed, B., Allison, S. D., Crenshaw, C., et al. (2008). Stoichiometry of soil enzyme activity at global scale. *Ecology Letters*, 11(11), 1252–1264. https://doi.org/10.1111/j.1461-0248.2008.01245.x
- Sinsabaugh, R. L., Manzoni, S., Moorhead, D. L., & Richter, A. (2013). Carbon use efficiency of microbial communities: Stoichiometry, methodology and modelling. *Ecology Letters*, 16(7), 930–939. https://doi.org/10.1111/j.1461-0248.2008.01245.x
- Six, J., Frey, S. D., Thiet, R. K., & Batten, K. M. (2006). Bacterial and fungal contributions to carbon sequestration in agroecosystems. Soil Science Society of America Journal, 70(2), 555–569. https://doi.org/10.2136/sssaj2004.0347
- Smith, P. (2004). Carbon sequestration in croplands: The potential in Europe and the global context. *European Journal of Agronomy*, 20, 229–236. https://doi.org/10.1016/j.eja.2003.08.002
- Smith, P., Smith, J. U., Powlson, D. S., McGill, W. B., Arah, J. R. M., Chertov, O. G., et al. (1997). A comparison of the performance of nine soil organic matter models using datasets from seven long-term experiments. *Geoderma*, 81(1–2), 153–225. https://doi.org/10.1016/ S0016-7061(97)00087-6
- Sulman, B. N., Moore, J. A., Abramoff, R., Averill, C., Kivlin, S., Georgiou, K., et al. (2018). Multiple models and experiments underscore large uncertainty in soil carbon dynamics. *Biogeochemistry*, 141(2), 109–123. https://doi.org/10.1007/s10533-018-0509-z
- Sulman, B. N., Phillips, R. P., Oishi, A. C., Shevliakova, E., & Pacala, S. W. (2014). Microbe-driven turnover offsets mineral-mediated storage of soil carbon under elevated CO₂. Nature Climate Change, 4(12), 1099–1102. https://doi.org/10.1038/nclimate2436
- Tarantola, S., & Becker, W. (2016). Simlab software for uncertainty and sensitivity analysis. In Springer handbook of uncertainty quantification (pp. 1–21). Springer. https://doi.org/10.1007/978-3-319-11259-6_61-1
- Van Wesemael, B., Paustian, K., Meersmans, J., Goidts, E., Barancikova, G., & Easter, M. (2010). Agricultural management explains historic changes in regional soil carbon stocks. Proceedings of the National Academy of Sciences, 107(33), 14926–14930. https://doi.org/10.1073/ pnas.1002592107
- Wang, C., Qu, L., Yang, L., Liu, D., Morrissey, E., Miao, R., et al. (2021). Large-scale importance of microbial carbon use efficiency and necromass to soil organic carbon. *Global Change Biology*, 27(10), 2039–2048. https://doi.org/10.1111/gcb.15550

Wang, G., Post, W. M., & Mayes, M. A. (2013). Development of microbial-enzyme-mediated decomposition model parameters through steady-state and dynamic analyses. *Ecological Applications*, 23(1), 255–272. https://doi.org/10.1890/12-0681.1

Wang, G., Post, W. M., Mayes, M. A., Frerichs, J. T., & Sindhu, J. (2012). Parameter estimation for models of ligninolytic and cellulolytic enzyme kinetics. Soil Biology and Biochemistry, 48, 28–38. https://doi.org/10.1016/j.soilbio.2012.01.011

Wieder, W. R., Allison, S. D., Davidson, E. A., Georgiou, K., Hararuk, O., He, Y., et al. (2015). Explicitly representing soil microbial processes in Earth system models. *Global Biogeochemical Cycles*, 29(10), 1782–1800. https://doi.org/10.1002/2015GB005188

Zhang, Y., Li, C., Trettin, C. C., Li, H., & Sun, G. (2002). An integrated model of soil, hydrology, and vegetation for carbon dynamics in wetland ecosystems. *Global Biogeochemical Cycles*, *16*(4), 1061. https://doi.org/10.1029/2001GB001838